

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Millimet, Daniel L.; Tchernis, Rusty

Working Paper
Anthropometric Mobility During Childhood

IZA Discussion Papers, No. 7453

Provided in Cooperation with:

IZA - Institute of Labor Economics

Suggested Citation: Millimet, Daniel L.; Tchernis, Rusty (2013): Anthropometric Mobility During Childhood, IZA Discussion Papers, No. 7453, Institute for the Study of Labor (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/80698

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





IZA DP No. 7453

Anthropometric Mobility During Childhood

Daniel L. Millimet Rusty Tchernis

June 2013

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Anthropometric Mobility During Childhood

Daniel L. Millimet

Southern Methodist University and IZA

Rusty Tchernis

Georgia State University, NBER and IZA

Discussion Paper No. 7453 June 2013

IZA

P.O. Box 7240 53072 Bonn Germany

Phone: +49-228-3894-0 Fax: +49-228-3894-180 E-mail: iza@iza.org

Any opinions expressed here are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but the institute itself takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent nonprofit organization supported by Deutsche Post Foundation. The center is associated with the University of Bonn and offers a stimulating research environment through its international network, workshops and conferences, data service, project support, research visits and doctoral program. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ABSTRACT

Anthropometric Mobility During Childhood*

While childhood obesity has become a significant public health concern over the last few decades, knowledge concerning the origins of or persistence in childhood anthropometric measures is incomplete. Here, we utilize several nonparametric measures of mobility to assess the evolution of weight, height, and body mass index during early childhood. We find that mobility is quite high *prior to* primary school and then declines noticeably. However, there are important sources of heterogeneity, including race, gender, and age, that should prove insightful to researchers and policymakers.

JEL Classification: C23, I12, I18

Keywords: childhood obesity, persistence, mobility

Corresponding author:

Rusty Tchernis
Department of Economics
Andrew Young School of Policy Studies
Georgia State University
Box 3992
Atlanta, GA 30302-3992
USA

E-mail: rtchernis@gsu.edu

_

This study was conducted by Georgia State University and Southern Methodist University under a cooperative agreement with the U.S. Department of Agriculture, Economic Research Service, Food and Nutrition Assistance Research Program (agreement no. 58-5000- 0-0080). The views expressed here are those of the authors and do not necessarily reflect those of the USDA or ERS. We are grateful for outstanding research assistance by Lorenzo Almada.

1 Introduction

The prevalence of overweight adolescents has tripled in the last thirty years; it has more than doubled for younger children. Defined as having an age- and sex-adjusted body mass index (BMI) above the 95th percentile, the prevalence of overweight children increased from 5% to 12.4% for 2-5 year old children and from 5% to 17.6% for 12 to 19 year-olds between 1976 and 2006 (Ogden et al. 2008). In addition, vast differences in the time trends of BMI increases have been documented: the incidence of overweight among white girls aged 12-19 has increased from 7.4% to 14.5% between 1988 and 2006, whereas the corresponding figures for African-American girls are 13.2% and 27.7% (Ogden et al. 2002; Ogden et al. 2008). Deckelbaum and Williams (2001, p. 242S) conclude that "childhood obesity is increasing at epidemic rates, even among pre-school children..." More recently, Brisbois et al. (2012, p. 347) state: "Obesity is considered to be a worldwide epidemic with little evidence that its incidence is declining or that it has even reached a plateau."

As childhood obesity has received greater attention, its consequences have becoming increasingly well-documented. Obesity burdens individuals with severe physical, economic, and emotional suffering, and puts children and adolescents at risk for a number of health problems such as those affecting cardiovascular health, the endocrine system, and mental health (Deckelbaum and Williams 2001; Krebs and Jacobson 2003). However, perhaps the largest cost of childhood obesity comes from its impact on adult obesity. Currently, 60% of the total U.S. population is overweight or obese and 50% is expected to be obese in 2030 at the current rate (Dor et al. 2010). Aside from the health costs of adult obesity, adult obesity results in lower wages, productivity, and self-esteem (Mocan and Tekin 2011). In the U.S., the total cost attributable to obesity was over \$75 billion in 2000 according to Finkelstein et al. (2004). More recent estimates put the cost over \$200 billion (Cawley and Meyerhoefer 2012).

While the changes in childhood and adult obesity rates across cohorts, as well as the consequences of these increases, are well-documented, much less is known about how anthropometric measures evolve over the life cycle for a given individual. How persistent is childhood obesity? Does this persistence vary by age, race, gender, or location? Are there periods of relatively greater mobility of children within the distributions of weight or height? These are fundamentally important questions for researchers as well as policymakers. If weight is persistent, then early intervention is preferable to waiting until adolescence or beyond. However, if persistence varies by age, then the optimal timing of policy interventions may be further refined such that child weight is reduced prior to the degree of persistence becoming elevated.

For instance, an article in the *New York Times* on March 22, 2010 states that some evidence now suggests that children may become entrenched "on an obesity trajectory" even before kindergarten; however, the evidence is not "ironclad." Public health officials tend to advocate school-based reforms in light of the near universal enrollment, yet others stress the importance of preschool interventions (e.g., Frisvold and Giri 2013; Dietz and Gortmaker 2001; Davis and Christoffel 1994). Eriksson et al. (2001, p. 735) conclude that "obesity is initiated early in life."

To shed light on these fundamental questions, we apply nonparametric measures of mobility to data from the Early Childhood Longitudinal Survey – Kindergarten Cohort (ECLS-K). The ECLS-K is a nationally representative longitudinal survey of children entering kindergarten in Fall 1998. In addition to providing information on birthweight,

¹See http://www.nytimes.com/2010/03/23/health/23obese.html.

anthropometric data is collected at several points in time between kindergarten and eighth grade. Thus, our focus here is on anthropometric mobility from infancy through adolescence. This spans two of the three 'critical' periods of development as it relates to obesity: the so-called age of adiposity rebound (roughly ages four to six) and the transition to adolescence (Dietz 1997).²

Our findings are striking. In particular, we obtain three key findings. First, while anthropometric mobility is relatively high over the period spanning birth to entry into kindergarten, weight, height, and BMI are highly persistent from kindergarten through middle school. However, there are periods of relatively greater mobility; anthropometric mobility is higher in periods corresponding to the two post-natal, critical periods posited in Dietz (1997). Specifically, we find that mobility is greatest at the start of kindergarten (typically age five) and during the transition from elementary to middle school (typically ages 11 to 14). As these periods coincide with movements to new schools for the majority of children, this suggests that interventions synced with transitions to new schools and new routines are likely to be more effective. That said, interventions prior to primary school are likely to be more effective (although not necessarily as cost-effective).

Second, heterogeneity is important. Consequently, a singular approach to the obesity epidemic is unlikely to be successful. Important sources of heterogeneity include: demographics, initial conditions, outcome measure, age range, and metric used to measure persistence. Third, non-whites and girls generally exhibit more anthropometric mobility and lower persistence prior to kindergarten entry; the reverse holds after kindergarten entry. As such, non-whites and female children are more likely to find themselves on an "obesity trajectory" earlier in life. The remainder of the paper is organized as follows. Section 2 provides a brief overview of the prior literature. Section 3 presents the empirical methodology and description of the data. Section 4 discusses the results. Section 5 concludes.

2 Related Literature

The persistence of childhood overweight status into adulthood has been documented in a number of studies. Whitaker et al. (1997) found that the probability of an overweight six year-old child becoming an obese adult is 50% compared to 10% for a non-overweight child. In addition, the risk of becoming obese in adulthood is exacerbated by having an obese parent. Eriksson et al. (2001) found that individuals were three times more likely to be obese as an adult if they had a BMI greater than 16, as opposed to below 14.5, at age seven. Nader et al. (2007) find that children who were overweight prior to age of five are five times as likely to be overweight at 12 relative to children who were not overweight prior to age of five.

Freedman et al. (2001) also report a strong relationship between overweight status in childhood and adult BMI. In a later study, Freedman et al. (2005) document significant differences in the transmission of BMI from childhood to adulthood along racial lines. The authors find that not only are overweight black children more likely to become obese adults, but also that "relatively thin (BMI \leq 50th percentile) white boys were more likely to become overweight adults than were their black counterparts" (p. 928). Thus, anthropometric mobility, both upward and downward, is greater for white children. Gable et al. (2008) analyze the relationship between socioeconomic status,

²The third 'critical' period represents the period of fetal development in utero.

overweight persistence, and school outcomes. The authors find that family socioeconomic status is predictive of both the probability of a child being overweight and the probability of persistence of overweight status. Van Cleave et al. (2010) find that prevalence of obesity is increasing and is highly persistent over time. Deckelbaum and Williams (2001, p. 239S) conclude: "Disturbingly, obesity in childhood, particularly in adolescence, is a key predictor for obesity in adulthood." Similarly, Dietz and Gortmaker (2001, p. 340) state: "The best evidence suggests that the majority of overweight adolescents go on to be overweight adults." Iughetti et al. (2008) provide an excellent summary of the literature on persistence in obesity.

In light of these relationships, Dietz and Robinson (2005, p. 2102) suggest that "treatment to achieve weight maintenance is recommended" for two to six year-old overweight children. In January 2010, the U.S. Preventive Services Task Force have issued new guidelines suggesting that doctors regularly screen the weight of children aged six and over and refer children to specialized weight management programs if needed.³

3 Empirics

3.1 Mobility Measures

To examine anthropometric mobility from birth through adolescence, we use several metrics commonly used to measure income mobility. Applying these metrics to different subgroups of the data, we are also able to analyze differences in mobility across demographic groups, as well as across stages of childhood.

To proceed, define the following notation. Let y_i^t , denote an anthropometric measure for child i, i = 1, ..., N, in period t, t = 1, ..., T. Further, let $F_{t_0,t_1}(y^{t_0}, y^{t_1})$ denote the joint (bivariate) cumulative distribution function (CDF) for two distinct periods, t_0 and t_1 , where $t_0 < t_1$ and $y^t \equiv [y_1^t \cdots y_N^t]$.

While movement through the distribution from an initial period, t_0 , to a subsequent period, t_1 , is completely captured by $F_{t_0,t_1}(y^{t_0},y^{t_1})$, this is not practical. One method by which to summarize this joint distribution is via a $K \times K$ transition matrix, P_{t_0,t_1} , with representative element

$$p_{kl}^{t_0,t_1} = \frac{\Pr(\zeta_{k-1}^{t_0} \le y^{t_0} < \zeta_k^{t_0}, \ \zeta_{l-1}^{t_1} \le y^{t_1} < \zeta_l^{t_1})}{\Pr(\zeta_{k-1}^{t_0} \le y^{t_0} < \zeta_k^{t_0})} \qquad k, l = 1, ..., K,$$

$$(1)$$

where $0 < \zeta_1^s < \zeta_2^s < \dots < \zeta_{K-1}^s < \infty$, $\zeta_0^s = 0$, and $\zeta_K^s = \infty$, $s = t_0, t_1$, are cutoff points between the K classes.⁴ Thus, $p_{kl}^{t_0,t_1}$ gives the fraction of children in class k in period t_0 who are in class l in period t_1 . Note, inclusion of the denominator in (1) standardizes elements of the transition matrix such that each row and column sums to unity. A complete lack of mobility implies $p_{kl}^{t_0,t_1}$ equals unity if k=l and zero otherwise. Finally, we can define conditional transition matrices computed using sub-samples with X=x, where X denotes a vector of individual attributes. Denote the conditional transition matrix as $P_{t_0,t_1}(x)$, with representative element

$$p_{kl}^{t_0,t_1}(x) = \frac{\Pr(\zeta_{k-1}^{t_0} \le y^{t_0} < \zeta_k^{t_0}, \ \zeta_{l-1}^{t_1} \le y^{t_1} < \zeta_l^{t_1} | X = x)}{\Pr(\zeta_{k-1}^{t_0} \le y^{t_0} < \zeta_k^{t_0} | X = x)} \qquad k, l = 1, ..., K.$$
(2)

Implicit in this definition is the assumption that X are time invariant attributes.

 $^{^3\}mathrm{See}\ \mathrm{http://www.ahrq.gov/clinic/uspstf/uspschobes.htm.}$

⁴For example, if K = 10, then the cutoff points might correspond to deciles within the two marginal distributions of y^{t_0} and y^{t_1} .

The shortcoming of transition matrices stems from the fact that they do not capture movements within classes. Recently, Bhattacharya and Mazumder (2011) present an alternative characterization of the joint distribution $F_{t_0,t_1}(y^{t_0},y^{t_1})$ to better assess upward and downward mobility. Their measure of upward mobility is defined as

$$v_{kl}^{t_0,t_1}(\delta) = \frac{\Pr(\zeta_{k-1}^{t_0} \le y^{t_0} < \zeta_k^{t_0}, \ F_{t_1}(y^{t_1}) - F_{t_0}(y^{t_0}) > \delta)}{\Pr(\zeta_{k-1}^{t_0} \le y^{t_0} < \zeta_k^{t_0})} \qquad k, l = 1, ..., K,$$

$$(3)$$

where $F_t(y^t)$ denotes the marginal distribution of y in period t and $\delta \in [0, 1 - F_0(\zeta_k^{t_0})]$ is a predefined constant representing the threshold defining upward mobility. In words, (3) captures the probability of an individual in the terminal period exceeding his or her initial percentile by at least δ , conditional on being located between $\zeta_{k-1}^{t_0}$ and $\zeta_k^{t_0}$ in the initial period. The corresponding measure of downward mobility is given by

$$\varpi_{kl}^{t_0,t_1}(\delta) = \frac{\Pr(\zeta_{k-1}^{t_0} \le y^{t_0} < \zeta_k^{t_0}, \ F_{t_1}(y^{t_1}) - F_{t_0}(y^{t_0}) < -\delta)}{\Pr(\zeta_{k-1}^{t_0} \le y^{t_0} < \zeta_k^{t_0})}, \tag{4}$$

where $\delta \in [0, F_{t_0}(\zeta_{k-1}^{t_0})]$. In words, (4) captures the probability of an individual in the terminal period reducing his or her initial percentile by at least δ , conditional on being located between $\zeta_{k-1}^{t_0}$ and $\zeta_k^{t_0}$ in the initial period. Finally, we can compute *conditional* measures of upward and downward mobility by conditioning on X = x as in (2). In the analysis, we set $\delta = 0.10$; mobility is defined as changing one's place in the distribution by at least ten percentile points.

While transition matrices, and the corresponding measures of upward and downward mobility, have the advantage of discretizing the continuous CDF, $F_{t_0,t_1}(y^{t_0},y^{t_1})$, they may not yield unambiguous rankings of the degree of mobility across different samples (e.g., comparing $v_{kl}^{t_0,t_1}$ across whites and non-whites for different combinations of k and l or comparing $v_{kl}^{t_0,t_1}$ with $v_{kl}^{t_1,t_2}$). Consequently, several *summary* measures, designed to capture the overall extent of mobility, have been proposed. Here, we analyze five such measures.

The first measure is due to Bartholomew (1982) and is a scalar computed using the elements of the transition matrix, P, with representative element given in (1). Formally, the measure is given by

$$M(P_{t_0,t_1}) = \frac{1}{K(K-2)} \sum_{k=1}^{K} \sum_{l=1}^{K} p_{kl}^{t_0,t_1} |k-l|.$$
 (5)

The Bartholomew (1982) measure captures the average number of classes crossed by all observations (Formby et al. 2004). As such, the measure takes into account not only whether children change class, but also the distance moved by children. However, the measure captures *relative* mobility only; absolute changes in anthropometric outcomes alone do not show up as mobility. The measure is normalized and is bounded between zero and one, with higher values representing greater mobility (Buchinsky and Hunt 1999).

The second measure is based on the correlation between anthropometric outcomes in periods t_0 and t_1 . Here, we use the Spearman rank correlation coefficient. Upon computation of the correlation coefficient, the measure of mobility is given by

$$M(\rho_{t_0,t_1}) = 1 - \rho_{t_0,t_1},\tag{6}$$

where ρ_{t_0,t_1} is the rank correlation. As with the first measure, the measure is bounded between zero and one with higher values representing greater mobility as ρ equals zero in both cases when the data are time independent (Fields

2000). Moreover, like the Bartholomew (1982) measure, the correlation-based measure also captures relative mobility only.

The next measure is based on the notion that, with greater mobility, outcomes aggregated over multiple periods should be more equal than outcomes from any individual period. Formally, the measure proposed in Shorrocks (1978) is given by

$$M_S = 1 - \frac{I(\overline{y})}{\frac{1}{t_1 - t_0 - 1} \sum_{t = t_0}^{t_1} \frac{\mu^t}{\mu^{t_0, t_1}} I(y^t)},$$
(7)

where $I(\cdot)$ is some measure of inequality, \overline{y} is a vector of anthropometric outcomes averaged over the periods t_0 to t_1 , μ^t is the mean outcome in period t, and μ^{t_0,t_1} is the mean outcome over the entire period spanning from t_0 to t_1 . The numerator of the ratio in (7) is the level of inequality in observation-specific mean outcomes, where the means are computed over the period t_0 to t_1 . The denominator represents a weighted average of inequality computed for each period from t_0 to t_1 . With a complete lack of mobility, the inequality of average outcomes is equal to the (weighted) average of per period inequality; thus, $M_S = 0$. With 'perfect' mobility, there is lots of inequality in any given period (i.e., the denominator in (7) is non-zero), but average outcomes are identical (i.e., the numerator is zero); thus, $M_S = 1$. Again, the measure is bounded between zero and one with higher values indicating greater mobility. It is also a relative measure only if the inequality measure is invariant to absolute changes that are rank-preserving. In the analysis, we use Theil's measure of inequality when computing (7), given by

$$I(x_1, ..., x_N) = \frac{1}{N} \sum_{i} \left[\frac{x_i}{\overline{x}} \cdot \ln \left(\frac{x_i}{\overline{x}} \right) \right],$$

where x represents some variable.

The final two measures come from Cowell and Flachaire (2011). Both measures can be expressed as

$$M_{CF}^{\alpha} = \begin{cases} \frac{1}{\alpha(\alpha - 1)} \left[\frac{\frac{1}{N} \sum_{i} \left(x_{i}^{t_{0}}\right)^{\alpha} \left(x_{i}^{t_{1}}\right)^{1 - \alpha}}{\left(\mu^{t_{0}}\right)^{\alpha} \left(\mu^{t_{1}}\right)^{1 - \alpha}} - 1 \right] & \alpha \neq 0, 1 \\ \frac{\frac{1}{N} \sum_{i} x_{i}^{t_{1}} \log \left(x_{i}^{t_{1}}\right) - \frac{1}{N} \sum_{i} x_{i}^{t_{1}} \log \left(x_{i}^{t_{0}}\right)}{\mu^{1}} + \log \left(\frac{\mu^{t_{0}}}{\mu^{t_{1}}}\right) & \alpha = 0 \\ \frac{\frac{1}{N} \sum_{i} x_{i}^{t_{0}} \log \left(x_{i}^{t_{0}}\right) - \frac{1}{N} \sum_{i} x_{i}^{t_{0}} \log \left(x_{i}^{t_{1}}\right)}{\mu^{t_{0}}} + \log \left(\frac{\mu^{t_{1}}}{\mu^{t_{0}}}\right) & \alpha = 1, \end{cases}$$

$$(8)$$

where a large, positive (negative) value of α produces an index that is particularly sensitive to downward (upward) movements. To operationalize M_{CF}^{α} for a given value of α , x^{t_0} and x^{t_1} must be defined. In the first case, we set $x_i^s = y_i^s$, s = 1, ..., T. In the second case, we set $x_i^s = F_s(y_i^s)$, where $F_s(\cdot)$ is the marginal CDF of y^s , s = 1, ..., T. Cowell and Flachaire (2011) refer to the second case as capturing rank (relative) mobility; the first case measures absolute and relative mobility. In practice, this second measure is implemented by replacing $F_s(\cdot)$ with its empirical counterpart, defined as

$$\widehat{F}_s(y) = \frac{1}{N} \sum_i \mathbf{1}(y_i^s \le y), \tag{9}$$

where $\mathbf{1}(\cdot)$ is the indicator function. In the analysis, we set $\alpha = 0$.

3.2 Data

We utilize data from the restricted version of the ECLS-K. Collected by the U.S. Department of Education, the ECLS-K surveys a nationally representative cohort of children throughout the U.S. in fall and spring kindergarten,

fall and spring first grade, spring third grade, spring fifth grade, and spring eighth grade. The sample includes data on over 20,000 students who entered kindergarten in roughly 1,000 schools during the 1998-99 school year. In addition to family background information, height and weight measures are available from children in each round, as well as information on birth weight.

Our final sample consists of children for whom we have valid measures of age, gender, height, and weight.⁵ From the information on height and weight of the children, we create z-scores for weight, height, and BMI. Since negative values create problems in the use of the measures in (7) and (8), we convert z-scores to percentiles. Note that z-scores and percentiles are based on CDC 2000 growth charts; these are age- and gender-specific, are adjusted for normal growth, and are based on the underlying reference population.⁶ Moreover, whereas percentiles for weight at birth are available, they are not for height and BMI. Finally, to assess heterogeneity in anthropometric mobility, we compute the various mobility measures for the full sample as well as subsamples defined on the basis of race (white vs. non-white), gender (male vs. female), and socioeconomic status (low vs. high SES).

4 Results

4.1 Percentile Weight

Tables 1-3 display the results for percentile weight. Table 1 reports the five summary measures of mobility; Tables 2 and 3 report the results for upward and downward mobility, respectively. In each table, Panel I contains the full sample results, Panels II and III contain the results disaggregated by race, Panels IV and V contain the results disaggregated by SES status. Within each panel, we report the mobility measures computed over several time periods, including the full sample period from birth through spring eighth grade, as well as the sub-periods from birth through fall kindergarten, fall kindergarten through spring eighth grade, fall kindergarten through spring first grade, spring first grade through spring third grade, spring third grade, and spring fifth grade through spring eighth grade. Finally, each table reports standard errors computed by simple, nonparametric bootstrap with 250 repetitions clustered at the child-level.

Turning to the summary measures of mobility in Table 1 for the full sample, two findings stand out. First, there exists a clear pattern: mobility follows a *U*-shaped pattern, being greatest between birth and kindergarten entry and lowest between spring third and fifth grades. For example, using Shorrocks' (1978) measure, mobility is 0.557 over the period spanning birth to fall kindergarten, 0.134 over the period spanning fall kindergarten to spring first grade, 0.108 over the period spanning spring first to spring third grade, 0.072 over the period spanning spring third to spring fifth grade, and 0.149 over the period spanning spring fifth to spring eighth grade. The estimates are very precise. Given the structure of schooling faced by most children in the U.S., this suggests that mobility is relatively high during infancy as well as during times of transition (i.e., at the start of primary school and then again at the start of

⁵The initial sample size of the ECLS-K is 21,260. After cleaning age, weight, and height as described in Millimet and Tchernis (2013, Appendix C), and due to sample attrition, the sample size falls to 9,360 in the final wave of the data. Restricting the sample to a balanced panel reduces the sample size to approximately 9,160. Restricting the sample to those with valid birth weight reduces the sample size per wave to roughly 8,370. Note, all sample sizes are rounded to the nearest 10 per NCES restricted data regulations.

⁶z-scores and their percentiles are obtained using the -zanthro- command in Stata.

middle school). This timing also coincides with two of the three 'critical' periods for the onset of childhood obesity; namely, the period of so-called adiposity rebound, typically occurring between ages four and six, and the transition into adolescence (Dietz 1997). In terms of policy advice, this suggests that interventions prior to kindergarten may have the greatest potential to influence future weight. Once children reach kindergarten age, mobility peaks during times of transition to a new school: kindergarten entry and the transition to middle school. Thus, interventions for school-age children have the potential to be more effective if they occur at a time when children are developing new routines in new school environments.

Second, the various measures generally agree as it relates to changes in the degree of mobility across time periods. However, the amount of mobility between fall kindergarten and spring first grade as compared with between spring fifth and eighth grade depends on the measure utilized. Whereas the Bartholomew (1982) and Shorrocks (1978) measures point to greater mobility during the transition to middle school, the other measures indicate greater mobility at the start of kindergarten.

Examining upward and downward mobility in Tables 2 and 3 permits a more detailed examination of mobility patterns. Specifically, the figures in the tables represent the probability of moving up or down by more than ten percentile points from the initial period to the terminal period conditional on the quintile the child belongs to in the initial period. In terms of upward mobility in Table 2, three noticeable patterns emerge. First, there is a negative, monotonic relationship across the quintiles when examining changes from birth through spring eighth grade, birth through fall kindergarten, and fall kindergarten through spring eighth grade. For example, the probability of moving up at least ten percentile points in the distribution from fall kindergarten to spring eighth grade is 0.457 for children initially in the first quintile in fall kindergarten, 0.398 for the second quintile, 0.314 for the third quintile, 0.194 for the fourth quintile, and 0.038 for the top quintile. While it is understandable that upward mobility is lowest for children initially in the fifth quintile (since only those below the 90th percentile have the ability to move up at least ten percentile points), there is nothing that guarantees that upward mobility must decline as one moves up the distribution in the initial period.

Second, despite upward mobility declining monotonically across the initial distribution when examining the period spanning fall kindergarten to spring eighth grade, when we assess upward mobility across different sub-periods during primary school, we find mobility is greatest for those initially in the second quintile. Finally, we continue to find strong evidence of a *U*-shaped relationship in mobility across time periods, consistent with the 'critical' periods of child development discussed earlier. In particular, upward mobility is lowest over the period spanning spring third to spring fifth grades. For example, the probability of moving up at least ten percentile points in the distribution for a child in fourth quintile in the initial period is 0.227 over the period from birth through kindergarten entry, 0.120 over the period spanning fall kindergarten to spring first grade, 0.112 over the period spanning spring first to spring third grade, 0.105 over the period spanning spring third to spring fifth grade, and 0.158 over the period spanning spring fifth to spring eighth grade. Thus, children in this category are about 50% more upwardly mobile during the transition to middle school than during the middle years of elementary school.

Examining downward mobility in Table 3, three patterns are noticeable. First, downward mobility is strictly increasing across the distribution over the periods spanning birth through spring eighth grade, birth through kinder-

garten entry, and fall kindergarten through spring eighth grade. However, there is an inverted U-shaped relationship across the quintiles, peaking at the fourth quintile, when examining mobility from fall kindergarten to spring eighth grade. For example, the probability of moving down at least ten percentile points in the distribution is 0.062 for children initially in the first quintile, 0.287 for the second quintile, 0.374 for the third quintile, 0.479 for the fourth quintile, and 0.390 for the top quintile. Second, despite downward mobility peaking at the fourth quintile when examining the period from fall kindergarten through spring eighth grade, downward mobility tends to be largest at the third quintile when we examine later sub-periods. Specifically, downward mobility is greatest for those initially in the fourth quintile during early primary school and greatest for those initially in the third quintile during later primary and middle school.

Finally, as with the prior mobility measures, we find strong evidence of a *U*-shaped relationship in downward mobility across time periods for most quintiles. In particular, for all quintiles downward mobility is lowest over the period spanning spring third to spring fifth grades. For example, the probability of moving down at least ten percentile points for a child in the highest quintile in the initial period is 0.700 over the period from birth through kindergarten entry, 0.108 over the period spanning fall kindergarten to spring first grade, 0.091 over the period spanning spring first to spring third grade, 0.067 over the period spanning spring third to spring fifth grade, and 0.140 over the period spanning spring fifth to spring eighth grade. Thus, children in this category are more than twice as downwardly mobile during the transition to middle school than during the middle years at elementary school. As stated above, the fact that mobility during primary school tends to be the greatest at the start of kindergarten and then over the period most likely spanning the start of middle school suggests that interventions coinciding with transitions in schooling, which also coincide with critical periods of biological development, are likely to have the greatest efficacy. However, mobility between birth and entry into kindergarten is much higher than during primary school suggesting the potential for greater impacts from interventions prior to elementary school.

Next, we turn to the results for different demographic groups. In the interest of brevity, we highlight a few salient findings. First, the patterns of mobility are quite distinct for whites and non-whites. All summary measures in Table 1 indicate greater mobility for non-whites over the period from birth to kindergarten entry and less mobility over the period from fall kindergarten through spring eighth grade. For example, the first Cowell and Flaichaire (2011) measure, CF(0), is 0.537 (0.417) for non-whites (whites) prior to kindergarten entry, but 0.133 (0.154) during primary school. Tables 2 and 3 provide greater insights into this pattern. Specifically, the greater mobility prior to kindergarten is driven predominantly by greater upward mobility by those initially in the first and second quintiles (Table 2). The lower mobility during elementary school holds across nearly the entire distribution as non-whites have lower upward and downward mobility over this time period.

Second, a similar pattern emerges when comparing boys to girls. All summary measures indicate *greater* mobility for girls prior to kindergarten; the reverse holds during primary school (Table 1). Tables 2 and 3 indicate that this is due to greater *upward* mobility by girls between birth and kindergarten entry, whereas boys have greater *upward* and *downward* mobility across the majority of quintiles during primary school. Finally, there is little definitive pattern to the results when we differentiate by SES status.

4.2 Percentile Height

Tables 4-6 present the analogous results for percentile height. Turning to the summary measures of mobility in Table 4 for the full sample, there exists a clear pattern that is distinct from that for weight: mobility is relatively constant over time between kindergarten and spring fifth grade. Mobility then roughly doubles over the period spanning spring fifth to eighth grade. For example, using Shorrocks' (1978) measure, mobility is 0.062 over the period spanning fall kindergarten to spring first grade, 0.072 over the period spanning spring first to spring third grade, 0.088 over the period spanning spring third to spring fifth grade, and 0.172 over the period spanning spring fifth to spring eighth grade. This spike in mobility during the transition to middle school coincides with the spike in weight mobility found in Table 1. As before, the estimates are very precise.

Examining upward and downward mobility in Tables 5 and 6 permits a more detailed examination of mobility patterns. In terms of upward mobility in Table 5, again two noticeable patterns emerge. First, there exists an inverted U-shaped relationship across the distribution within any given time period. For example, upward mobility over the period spanning spring fifth to spring eighth grade is 0.304 for the first quintile, 0.404 for the second quintile, 0.355 for the third quintile, 0.260 for the fourth quintile, and 0.044 for the fifth quintile. Second, as in Table 4, upward mobility tends to be relatively constant across time periods within a quintile until peaking during the period from spring fifth to spring eighth grade. For instance, for the fourth quintile, the probability of moving up at least ten percentile points in the distribution is 0.122 over the period spanning fall kindergarten to spring first grade, 0.104 over the period spanning spring first to spring third grade, 0.136 over the period spanning spring third to spring fifth grade, and 0.260 over the period spanning spring fifth to spring eighth grade. Thus, children in this category are about 50% more upwardly mobile during the transition to middle school than during the middle years of elementary school.

Examining downward mobility in Table 6, we point to two main findings. First, downward mobility is strictly increasing across the distribution over the periods spanning fall kindergarten through spring eighth grade. However, there is an inverted *U*-shaped relationship across the quintiles, peaking at the fourth quintile, when examining downward mobility over the earlier sub-periods in the sample. For example, the probability of moving down at least ten percentile points in the distribution over the period spanned by spring third to spring fifth grade is 0.011 for children initially in the first quintile, 0.131 for the second quintile, 0.212 for the third quintile, 0.215 for the fourth quintile, and 0.118 for the top quintile. Second, as with upward mobility, downward mobility tends to be relatively constant across time periods within a quintile until peaking during the period from spring fifth to spring eighth grade. For instance, for the fourth quintile, the probability of moving down at least ten percentile points in the distribution is 0.185 over the period spanning fall kindergarten to spring first grade, 0.178 over the period spanning spring first to spring fifth to spring eighth grade.

Next, we turn to the results for different demographic groups. In the interest of continued brevity, we focus on a few key results. First, there is little difference in the summary mobility measures, or upward and downward mobility, along racial lines over the period spanning fall kindergarten to spring eighth grade. However, the summary mobility measures in every sub-period, and upward and downward mobility in many sub-periods for many quintiles, are higher for non-whites. This pattern could materialize for two reasons. On the one hand, (relative) height may be much more volatile over short intervals for non-whites, although not over long intervals. On the other hand, height may be more noisy from one period to the next due to greater measurement error.

Second, there are large differences in mobility along gender lines. Specifically, mobility over the period from spring fifth to spring eighth grade is much greater for girls. For example, the summary measures are about 50-80% higher for girls over this time period (Table 4). This is driven mainly by differential upward mobility, which is about 10 percentage points higher for girls across the bottom four quintiles (Table 5). However, girls initially in the top quintile as of spring fifth grade are also much more likely to experience downward mobility between spring fifth and spring eighth grade (Table 6). The greater mobility for girls over this time period is presumably due to variation in the onset of puberty, which may begin earlier for girls than boys (Herman-Giddens 2006). It is also consistent with the period of transition into adolescence being of greater biological significance for girls' height. Finally, as with weight, there is little definitive pattern to the results when we differentiate by SES status.

4.3 Percentile BMI

Tables 7-9 present the final set of results for percentile BMI. Mobility in BMI reflects differential mobility of weight and height and is of more direct relevance due to the costs mentioned previously associated with obesity. Turning to the summary measures of mobility in Table 7 for the full sample, two findings stand out. First, as with weight, mobility follows a U-shaped pattern. For example, using Shorrocks' (1978) measure, mobility is 0.437 over the period spanning fall kindergarten to spring first grade, 0.313 over the period spanning spring first to spring third grade, 0.170 over the period spanning spring third to spring fifth grade, and 0.230 over the period spanning spring fifth to spring eighth grade. However, unlike with weight, all measures agree that mobility is greater over the period spanning fall kindergarten to spring first grade than over the period spanning spring fifth grade to spring eighth grade. Second, there is greater mobility in terms of BMI than for either weight or height alone. For example, over the period spanning kindergarten entry to spring eighth grade, the first Cowell and Flaichaire (2011) measure, CF(0), is 0.221 for BMI, but only 0.146 and 0.111 for weight and height, respectively.

Examining upward mobility in Table 8, we notice two patterns. First, as with weight, there is a negative, monotonic relationship across the quintiles when examining changes from fall kindergarten through spring eighth grade, as well as during the first two sub-periods. For example, the probability of moving up at least ten percentile points in the distribution from fall kindergarten to spring eighth grade is 0.543 for children initially in the first quintile in fall kindergarten, 0.409 for the second quintile, 0.311 for the third quintile, 0.189 for the fourth quintile, and 0.032 for the top quintile. Upward mobility is greatest for those initially in the second quintile for the final two sub-periods. Second, we continue to find strong evidence of a *U*-shaped relationship in mobility across time periods, consistent with the 'critical' periods of child development discussed earlier. In particular, upward mobility is lowest over the period spanning spring third to spring fifth grades. Moreover, upward mobility is roughly equal in the first and final sub-periods with the exception of the first quintile; here, mobility is greatest between fall kindergarten and spring first grade.

Turning to downward mobility in Table 9, three patterns are noticeable. First, as with weight, downward mobility

follows an inverted U-shaped relationship across the quintiles, peaking at the fourth quintile, when examining mobility from fall kindergarten to spring eighth grade. For example, the probability of moving down at least ten percentile points in the distribution is 0.052 for children initially in the first quintile, 0.311 for the second quintile, 0.417 for the third quintile, 0.513 for the fourth quintile, and 0.410 for the top quintile. Second, again consistent with weight, downward mobility tends to be largest at the third quintile when we examine later sub-periods. Finally, as with the prior mobility measures, we find strong evidence of a U-shaped relationship in downward mobility across time periods for most quintiles. In particular, for all quintiles downward mobility is lowest over the period spanning spring third to spring fifth grades. For example, the probability of moving down at least ten percentile points for a child in the highest quintile in the initial period is 0.186 over the period spanning fall kindergarten to spring first grade, 0.117 over the period spanning spring first to spring third grade, 0.090 over the period spanning spring third to spring fifth grade, and 0.142 over the period spanning spring fifth to spring eighth grade.

Lastly, we turn to the results for different demographic groups. When splitting the sample along racial lines, all summary measures in Table 7, except Shorrocks (1978) measure, indicate greater mobility for whites over the period from fall kindergarten through spring eighth grade. Specifically, the greater mobility is driven predominantly by greater upward mobility throughout the distribution, becoming somewhat more pronounced during the transition to middle school (Table 2). Whites also experience greater downward mobility, but here the effects tend to be concentrated at the upper quintiles.

The differences are not so stark when we split the sample by gender. Now, while all summary measures indicate greater mobility for boys over the full period spanning kindergarten entry to spring eighth grade, the results vary by sub-period. Boys (Girls) are more mobile according to all measures over the first (last) two sub-periods. This is consistent with the period of adiposity rebound being more critical for determining future BMI for boys, whereas the transition into adolescence is more critical for girls. Tables 8 and 9 indicate that this mobility pattern is due primarily to greater downward mobility by boys over the first two sub-periods and greater upward and downward mobility by girls over the last two sub-periods. As with weight and height, there is little definitive pattern to the results when we differentiate by SES status.

5 Conclusion

Concern over childhood obesity has risen dramatically over the past decade. However, our knowledge has not kept pace with this concern. In particular, our knowledge over how health evolves over time is sorely lacking. Our analysis here borrows several commonly used nonparametric measures of mobility to assess anthropometric mobility during early childhood using longitudinal data from the ECLS-K.

We arrive at three main takeaway points. First, while there is significant mobility in terms of weight between birth and the start of primary school, there is much less mobility once children are of primary school age. The greater mobility over the period spanning birth to kindergarten entry is consistent with the notion that birth weight alone is not a great indicator of future weight status. As noted in the medical literature, superior measures must account for weight relative to gestation age, head circumference, and other anthropometric indicators (e.g., Brisbois et al. 2012).

Second, heterogeneity is very important. There is no universal measure of persistence or mobility; no single measure can capture the complex movements that occur throughout a given distribution over time. The analysis points to three primary sources of heterogeneity: initial quintile, age range, and demographic group. By initial quintile we mean that where a child starts in the distribution leads to differences in mobility patterns even conditional on the time period over which one is assessing mobility and conditional on the demographic group to which the child belongs. By age range we mean that mobility patterns vary depending on the time period one is assessing mobility, even holding constant a child's initial place in the distribution and demographic group. Finally, by demographic group, we mean that mobility patterns vary across demographic groups, even conditional on a child's initial place in the distribution and the time period. As discussed above, perhaps the most intriguing finding is the relative amounts of mobility occurring at the start of primary school (i.e., between fall kindergarten and spring first grade) and between the final two waves in the data (i.e., spring fifth to spring eighth grade). Thus, interventions aimed at encouraging new, healthy behaviors are perhaps most likely to be effective when implemented during key transition periods such as entry into elementary and middle school, particularly when those transitions also coincide with evidence from the medical literature indicating these as critical periods of biological development. That said, when examining weight alone, mobility is significantly greater over the period spanned by birth to kindergarten entry. Thus, while interventions may be more difficult to implement during this time period, relative to school-based reforms during periods with near universal enrollment, they are likely to be more effective.

Third, the patterns of mobility are quite distinct along racial and gender lines; surprisingly, the differences are less systematic when we differentiate children by SES status. We find that whites and boys experience greater anthropometric mobility during primary school (i.e., the full period spanning kindergarten entry through the end of middle school). However, non-whites and girls experience greater anthropometric mobility prior to primary school. Girls also experience greater mobility during the transition to adolescence despite having lower mobility overall during primary school. This suggests the potential for non-whites, and girls to a lesser extent, to be on an "obesity trajectory" earlier in life.

In light of the results, future work is needed to explore the sources of anthropometric mobility, or the lack thereof, during different time periods and across different demographic groups.

References

- [1] Bartholomew, D.J. (1982), Stochastic Models for Social Processes, 3rd edition, Wiley: London.
- [2] Bhattacharya, D. and B. Mazumder (2011), "A Nonparametric Analysis of Black-White Differences in Intergenerational Mobility in the United States," *Quantitative Economics*, 2, 335-379.
- [3] Brisbois, T.D., Farmer, A.P., and L. J. McCargar (2012), "Early Markers of Adult Obesity: A Review," Obesity Reviews, 13, 347-367.
- [4] Buchinsky, M., and J. Hunt (1999), "Wage Mobility in the United States," Review of Economics and Statistics, 81(3), 351-368.
- [5] Cawley, J. and C. Meyerhoefer (2012), "The Medical Care Costs of Obesity: An Instrumental Variables Approach," Journal of Health Economics, 31, 219-230.
- [6] Cowell, F.A. and E. Flachaire (2011), "Measuring Mobility," ECINEQ WP 2011-199.
- [7] Davis, K. and K.K. Christoffel (1994), "Obesity in Preschool and School-Age Children: Treatment Early and Often May Be Best," Archives of Pediatric and Adolescent Medicine, 148, 1257-1261.
- [8] Deckelbaum, R., and C. Williams (2001), "Childhood Obesity: The Health Issue," Obesity Research, 9, 239Sâ€"243S.
- [9] Dietz, W.H. (1997), "Periods of Risk in Childhood for the Development of Adult Obesity—What Do We Need to Learn?" Journal of Nutrition, 127, 1884S-1886S
- [10] Dietz, W.H., and S.L. Gortmaker (2001), "Preventing obesity in children and adolescents," Annual Review of Public Health, 22, 337-353.
- [11] Dietz, W.H. and T.N. Robinson (2005), "Overweight Children and Adolescents," New England Journal of Medicine, 352, 2100–2109.
- [12] Dor, A., C. Ferguson, E. Tan, C. Langwith (2010), "A Heavy Burden: The Individual Costs of Being Overweight and Obese in the United States," Department of Health Policy, George Washington University.
- [13] Eriksson, J., T. Forsén, J. Tuomilehto, C. Osmond, and D. Barker (2001), "Size at Birth, Childhood Growth and Obesity in Adult Life," *International Journal of Obesity*, 25, 735-740.
- [14] Fields, G.S. (2000), "Income Mobility: Concepts and Measures," in N. Birdsall and C. Graham (eds.) New Markets, New Opportunities? Economic and Social Mobility in a Changing World, Brookings Institution and the Carnegie Endowment.
- [15] Finkelstein, E. A., I.C. Fiebelkorn, and G. Wang (2004), "State-Level Estimates of Annual Medical Expenditures Attributable to Obesity," *Obesity Research*, 12.

- [16] Freedman, D.S., L.K. Khan, W.H. Dietz, S.R. Srinivasan, and G.S. Berenson (2001), "Relationship of Childhood Obesity to Coronary Heart Disease Risk Factors in Adulthood: The Bogalusa Heart Study," *Pediatrics*, 108, 712–718.
- [17] Formby, J., Smith, W., and B. Zheng (2004), "Mobility Measurement, Transition Matrices and Statistical Inference," Journal of Economic Inequality, 8, 409-427.
- [18] Freedman, D.S., L.K. Khan, M.K. Serdula, W.H. Dietz, S.R. Srinivasan, and G.S. Berenson (2005), "Racial Differences in the Tracking of Childhood BMI to Adulthood," Obesity Research, 13, 928–935.
- [19] Frisvold, D.E. and A. Giri (2013), "The Potential of Early Childhood Education as a Successful Obesity Intervention," in V. Brennan, S. Kumanyika, and R. Zambrana (eds.), Obesity Interventions in Underserved US Populations: Evidence and Directions, Johns Hopkins University Press.
- [20] Gable, S., J. Britt-Rankin, and J. Krull (2008), "Ecological Predictors and Developmental Outcomes of Persistent Childhood Overweight," Contractor and Cooperator Report No. 42, USDA, ERS.
- [21] Herman-Giddens, M.E. (2006), "Recent Data on Pubertal Milestones in United States Children: The Secular Trend Toward Earlier Development," *International Journal of Andrology*, 29, 241-246.
- [22] Iughetti, L., M. de Simone, A. Verrotti, M.L. Iezzi, B. Predieri, P. Bruzzi, S. Bernasconi, F. Balli, and G. Bedogni (2008), "Thirty-year Persistence of Obesity after Presentation to a Pediatric Obesity Clinic," Annals of Human Biology, 35, 439-448
- [23] Krebs, N.F., and M.S. Jacobson (2003), "Prevention of Pediatric Overweight and Obesity," *Pediatrics*, 112, 424–430.
- [24] Millimet, D.L. and R. Tchernis (2013), "Estimating Treatment Effects Without an Exclusion Restriction: With an Application to the School Breakfast Program," *Journal of Applied Econometrics*, forthcoming.
- [25] Mocan, N. and E. Tekin (2011), "Obesity, Self-Esteem, and Wages," in M. Grossman and N. Mocan (eds.) Economic Aspects of Obesity, University of Chicago Press.
- [26] Nader P.R., M. O'Brien, R. Houts, R. Bradley, J. Belsky, R. Crosnoe, S. Friedman, Z. Mei, and E.J. Susman (2007), "Identifying the Risk for Obesity in Early Childhood," *Pediatrics*, 118, e594–e601.
- [27] Ogden, C.L., K.M. Flegal, M.D. Carroll, and C.L. Johnson (2002), "Prevalence and Trends in Overweight among US Children and Adolescents, 1999-2000," *Journal of the American Medical Association*, 288(14), 1728–1732.
- [28] Ogden, C.L., M.D. Carroll, and K.M. Flegal, (2008), "High Body Mass Index for Age Among US Children and Adolescents," *Journal of the American Medical Association*, 299, 2401–2405.
- [29] Shorrocks, A.F. (1978), "Income Inequality and Income Mobility," Journal of Economic Theory, 19, 376-393.
- [30] Van Cleave, J., S.L. Gortmaker and J.M. Perrin (2010), "Dynamics of Obesity and Chronic Health Conditions Among Children and Youth," Journal of the American Medical Association, 303 (7), 623–630.

[31] Whitaker, R.C., J.A. Wright, M.S. Pepe, K.D. Seidel, and W.H. Dietz (1997), "Predicting Obesity in Young Adulthood from Childhood and Parental Obesity," *New England Journal of Medicine*, 337, 869-873.

Table 1. Mobility Measures: Percentile Weight

	M(P)	Μ(ρ)	M_{S}	CF(0)	CF _R (0)		M(P)	Μ(ρ)	M_{S}	CF(0)	CF _R (0)
					H. /		` ′				
I. Full Sa	ample										
[0,7]	0.334	0.853	0.458	0.506	0.437						
	(0.003)	(0.011)	(0.006)	(0.012)	(0.007)						
[0,1]	0.312	0.763	0.557	0.467	0.378						
	(0.003)	(0.010)	(0.007)	(0.011)	(0.006)						
[1,7]	0.192	0.321	0.271	0.146	0.175						
	(0.002)	(0.007)	(0.005)	(0.005)	(0.005)						
[1,4]	0.102	0.111	0.134	0.056	0.063						
	(0.002)	(0.003)	(0.005)	(0.003)	(0.002)						
[4,5]	0.097	0.093	0.108	0.040	0.051						
	(0.001)	(0.003)	(0.006)	(0.002)	(0.002)						
[5,6]	0.080	0.060	0.072	0.024	0.032						
	(0.001)	(0.002)	(0.003)	(0.001)	(0.001)						
[6,7]	0.110	0.106	0.149	0.041	0.054						
	(0.001)	(0.003)	(0.004)	(0.001)	(0.001)						
II. White							n-White				
[0,7]	0.331	0.840	0.461	0.463	0.431	[0,7]	0.334	0.850	0.448	0.564	0.433
	(0.003)	(0.013)	(0.006)	(0.015)	(0.008)		(0.004)	(0.017)	(0.009)	(0.022)	(0.010)
$[0,\!1]$	0.308	0.747	0.552	0.417	0.367	[0,1]	0.314	0.771	0.558	0.537	0.387
	(0.003)	(0.013)	(0.010)	(0.014)	(0.008)		(0.004)	(0.017)	(0.012)	(0.021)	(0.009)
[1,7]	0.201	0.347	0.273	0.154	0.189	[1,7]	0.180	0.280	0.266	0.133	0.152
	(0.003)	(0.009)	(0.006)	(0.007)	(0.006)		(0.003)	(0.010)	(0.008)	(0.007)	(0.006)
[1,4]	0.106	0.118	0.134	0.059	0.067	[1,4]	0.096	0.100	0.134	0.053	0.057
	(0.002)	(0.004)	(0.005)	(0.003)	(0.003)		(0.002)	(0.005)	(0.008)	(0.004)	(0.004)
[4,5]	0.101	0.103	0.107	0.040	0.054	[4,5]	0.090	0.078	0.109	0.041	0.044
	(0.002)	(0.003)	(0.007)	(0.002)	(0.002)		(0.002)	(0.004)	(0.012)	(0.005)	(0.003)
[5,6]	0.083	0.064	0.070	0.023	0.032	[5,6]	0.076	0.054	0.076	0.025	0.030
	(0.001)	(0.002)	(0.003)	(0.001)	(0.001)		(0.002)	(0.003)	(0.006)	(0.003)	(0.003)
[6,7]	0.114	0.115	0.157	0.045	0.058	[6,7]	0.102	0.091	0.136	0.035	0.045
	(0.002)	(0.004)	(0.005)	(0.002)	(0.002)		(0.002)	(0.004)	(0.007)	(0.002)	(0.002)

Notes: Period 0 = birth; Period 1 = fall kindergarten; Period 4 = spring 1st grade; Period 5 = spring 3rd grade; Period 6 = spring 5th grade; and, Period 7 = spring 8th grade. Bootstrap standard errors based on 250 repetitions (clustered at the child-level) in parentheses. N = 8370 (full sample), 5330 (white sample), 3040 (non-white sample), 4230 (male sample), 4140 (female sample), 2660 (low SES sample), and 5710 (high SES sample) with sample sizes rounded to the nearest 10 per NCES restricted data regulations. See text for definition of mobility measures.

Table 1 (cont.). Mobility Measures: Percentile Weight

	M(P)	Μ(ρ)	M_{S}	CF(0)	CF _R (0)		M(P)	Μ(ρ)	$M_{\rm S}$	CF(0)	CF _R (0)
											_
IV. Male	!					V. Fen	nale				
[0,7]	0.340	0.870	0.464	0.493	0.442	[0,7]	0.328	0.833	0.453	0.519	0.429
	(0.004)	(0.015)	(0.007)	(0.015)	(0.008)		(0.004)	(0.016)	(0.007)	(0.018)	(0.009)
[0,1]	0.309	0.755	0.544	0.435	0.371	[0,1]	0.315	0.771	0.570	0.499	0.383
	(0.004)	(0.014)	(0.011)	(0.014)	(0.008)		(0.004)	(0.015)	(0.011)	(0.020)	(0.009)
[1,7]	0.199	0.339	0.284	0.156	0.184	[1,7]	0.184	0.298	0.257	0.135	0.164
	(0.003)	(0.010)	(0.008)	(0.007)	(0.006)		(0.003)	(0.010)	(0.007)	(0.007)	(0.006)
[1,4]	0.105	0.119	0.144	0.060	0.067	[1,4]	0.098	0.103	0.124	0.052	0.059
	(0.002)	(0.005)	(0.007)	(0.004)	(0.003)		(0.002)	(0.004)	(0.006)	(0.003)	(0.003)
[4,5]	0.100	0.100	0.121	0.043	0.054	[4,5]	0.095	0.086	0.096	0.038	0.046
	(0.002)	(0.004)	(0.010)	(0.004)	(0.003)		(0.002)	(0.004)	(0.008)	(0.003)	(0.003)
[5,6]	0.078	0.059	0.071	0.020	0.031	[5,6]	0.081	0.061	0.074	0.027	0.033
	(0.002)	(0.003)	(0.003)	(0.001)	(0.001)		(0.002)	(0.002)	(0.004)	(0.002)	(0.002)
[6,7]	0.106	0.100	0.149	0.038	0.051	[6,7]	0.112	0.110	0.148	0.045	0.054
	(0.002)	(0.003)	(0.005)	(0.002)	(0.002)		(0.002)	(0.004)	(0.006)	(0.002)	(0.002)
VI. Low	SES					VII. H	igh SES				
[0,7]	0.328	0.827	0.464	0.543	0.426	[0,7]	0.334	0.854	0.454	0.482	0.435
. , .	(0.005)	(0.018)	(0.009)	(0.023)	(0.011)	. , .	(0.003)	(0.013)	(0.006)	(0.015)	(0.008)
[0,1]	0.306	0.742	0.566	0.510	0.372	[0,1]	0.314	0.771	0.552	0.444	0.380
- / -	(0.005)	(0.018)	(0.012)	(0.021)	(0.010)	- / -	(0.003)	(0.013)	(0.010)	(0.014)	(0.008)
[1,7]	0.188	0.306	0.291	0.153	0.166	[1,7]	0.193	0.324	0.260	0.142	0.177
- / -	(0.004)	(0.012)	(0.010)	(0.009)	(0.008)	- / -	(0.003)	(0.009)	(0.006)	(0.006)	(0.006)
[1,4]	0.099	0.107	0.142	0.056	0.059	[1,4]	0.103	0.113	0.130	0.056	0.065
	(0.003)	(0.006)	(0.009)	(0.004)	(0.003)		(0.002)	(0.004)	(0.005)	(0.003)	(0.003)
[4,5]	0.095	0.087	0.125	0.049	0.049	[4,5]	0.098	0.096	0.099	0.036	0.050
	(0.002)	(0.004)	(0.014)	(0.006)	(0.003)		(0.002)	(0.004)	(0.006)	(0.002)	(0.002)
[5,6]	0.076	0.055	0.083	0.026	0.031	[5,6]	0.081	0.063	0.067	0.022	0.032
•	(0.002)	(0.003)	(0.006)	(0.003)	(0.003)		(0.001)	(0.002)	(0.003)	(0.001)	(0.001)
[6,7]	0.108	0.100	0.155	0.040	0.051	[6,7]	0.111	0.110	0.148	0.042	0.055
	(0.003)	(0.005)	(0.009)	(0.002)	(0.003)		(0.002)	(0.003)	(0.005)	(0.001)	(0.002)
	,	ŕ	ŕ	. ,				ŕ	ŕ	ŕ	,

Table 2. Upward Mobility Measures: Percentile Weight

	Q1	Q2	Q3	Q4	Q5		Q1	Q2	Q3	Q4	Q5
I. Full Sa	ımple										
[0,7]	0.727	0.538	0.369	0.215	0.037						
	(0.010)	(0.011)	(0.011)	(0.010)	(0.005)						
[0,1]	0.680	0.530	0.388	0.227	0.034						
	(0.010)	(0.011)	(0.012)	(0.010)	(0.004)						
[1,7]	0.457	0.398	0.314	0.194	0.038						
	(0.011)	(0.012)	(0.011)	(0.009)	(0.004)						
[1,4]	0.217	0.225	0.222	0.120	0.008						
	(0.009)	(0.009)	(0.009)	(0.007)	(0.002)						
[4,5]	0.186	0.260	0.209	0.112	0.011						
	(0.008)	(0.010)	(0.009)	(0.007)	(0.003)						
[5,6]	0.146	0.227	0.192	0.105	0.006						
	(0.008)	(0.010)	(0.009)	(0.008)	(0.002)						
[6,7]	0.247	0.325	0.251	0.158	0.018						
	(0.009)	(0.011)	(0.010)	(0.009)	(0.003)						
TT \\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\						III No.	- XX/l-:4.a				
II. White		0.520	0.262	0.214	0.026		n-White	0.540	0.205	0.100	0.025
[0,7]	0.727	0.538	0.362	0.214	0.026	[0,7]	0.731	0.548	0.385	0.190	0.025
ΓΩ 11	(0.013)	(0.015)	(0.014)	(0.011)	(0.006)	ΓΩ 11	(0.017)	(0.019)	(0.019)	(0.016)	(0.007)
[0,1]	0.663	0.533	0.387	0.233	0.041	[0,1]	0.691	0.551	0.364	0.211	0.029
[4 5]	(0.013)	(0.015)	(0.015)	(0.012)	(0.006)	[4 8]	(0.018)	(0.019)	(0.018)	(0.015)	(0.008)
[1,7]	0.472	0.404	0.313	0.194	0.042	[1,7]	0.429	0.387	0.309	0.191	0.033
F4 41	(0.013)	(0.015)	(0.012)	(0.011)	(0.006)	F4 41	(0.016)	(0.018)	(0.017)	(0.015)	(0.007)
[1,4]	0.225	0.235	0.221	0.127	0.012	[1,4]	0.202	0.217	0.208	0.095	0.012
F 4 = 7	(0.011)	(0.013)	(0.012)	(0.009)	(0.004)	F 4 5 3	(0.014)	(0.018)	(0.016)	(0.013)	(0.005)
[4,5]	0.186	0.269	0.209	0.130	0.011	[4,5]	0.182	0.259	0.188	0.103	0.005
F. C.	(0.011)	(0.012)	(0.011)	(0.010)	(0.003)	F. 63	(0.015)	(0.017)	(0.015)	(0.013)	(0.003)
[5,6]	0.146	0.241	0.204	0.108	0.010	[5,6]	0.148	0.220	0.171	0.100	0.000
F C = 3	(0.010)	(0.013)	(0.012)	(0.011)	(0.004)	r < = 1	(0.013)	(0.016)	(0.015)	(0.012)	(0.002)
[6,7]	0.254	0.317	0.267	0.177	0.015	[6,7]	0.233	0.296	0.238	0.126	0.007
	(0.012)	(0.014)	(0.014)	(0.011)	(0.005)		(0.016)	(0.019)	(0.016)	(0.013)	(0.006)

Notes: Period 0 = birth; Period 1 = fall kindergarten; Period 4 = spring 1st grade; Period 5 = spring 3rd grade; Period 6 = spring 5th grade; and, Period 7 = spring 8th grade. Q1 - Q5 refers to the first through fifth quantiles in the initial period. Bootstrap standard errors based on 250 repetitions (clustered at the child-level) in parentheses. N = 8370 (full sample), 5330 (white sample), 3040 (non-white sample), 4230 (male sample), 4140 (female sample), 2660 (low SES sample), and 5710 (high SES sample) with sample sizes rounded to the nearest 10 per NCES restricted data regulations. See text for definition of mobility measures ($\delta = 0.10$).

Table 2 (cont.). Upward Mobility Measures: Percentile Weight

	Q1	Q2	Q3	Q4	Q5		Q1	Q2	Q3	Q4	Q5
IV. Male						V. Fen					
[0,7]	0.737	0.544	0.372	0.208	0.032	[0,7]	0.702	0.512	0.351	0.211	0.026
	(0.014)	(0.016)	(0.016)	(0.012)	(0.007)		(0.013)	(0.016)	(0.015)	(0.015)	(0.006)
$[0,\!1]$	0.677	0.519	0.375	0.220	0.033	[0,1]	0.668	0.532	0.379	0.231	0.041
	(0.015)	(0.016)	(0.018)	(0.014)	(0.006)		(0.015)	(0.016)	(0.016)	(0.017)	(0.008)
[1,7]	0.479	0.410	0.312	0.204	0.041	[1,7]	0.432	0.392	0.314	0.188	0.028
	(0.015)	(0.016)	(0.015)	(0.013)	(0.007)		(0.016)	(0.016)	(0.018)	(0.012)	(0.006)
[1,4]	0.222	0.227	0.232	0.131	0.011	[1,4]	0.218	0.223	0.216	0.107	0.006
	(0.013)	(0.015)	(0.014)	(0.013)	(0.003)		(0.013)	(0.013)	(0.013)	(0.010)	(0.003)
[4,5]	0.185	0.268	0.216	0.111	0.010	[4,5]	0.171	0.263	0.205	0.120	0.011
	(0.012)	(0.014)	(0.013)	(0.010)	(0.004)		(0.013)	(0.015)	(0.013)	(0.011)	(0.005)
[5,6]	0.149	0.240	0.167	0.104	0.008	[5,6]	0.158	0.211	0.187	0.114	0.002
	(0.011)	(0.014)	(0.013)	(0.010)	(0.003)		(0.011)	(0.014)	(0.013)	(0.012)	(0.003)
[6,7]	0.244	0.301	0.247	0.139	0.012	[6,7]	0.223	0.331	0.265	0.176	0.013
	(0.015)	(0.015)	(0.015)	(0.013)	(0.004)		(0.013)	(0.015)	(0.014)	(0.013)	(0.004)
VI. Low	SES					VII. H	igh SES				
[0,7]	0.737	0.509	0.368	0.224	0.029	[0,7]	0.732	0.540	0.377	0.204	0.035
	(0.017)	(0.020)	(0.021)	(0.017)	(0.008)		(0.013)	(0.014)	(0.014)	(0.010)	(0.006)
[0,1]	0.690	0.513	0.382	0.239	0.038	[0,1]	0.678	0.542	0.398	0.216	0.042
	(0.017)	(0.021)	(0.020)	(0.019)	(0.008)		(0.013)	(0.013)	(0.015)	(0.012)	(0.006)
[1,7]	0.433	0.397	0.326	0.190	0.045	[1,7]	0.453	0.405	0.317	0.206	0.040
	(0.017)	(0.019)	(0.018)	(0.015)	(0.009)		(0.012)	(0.014)	(0.014)	(0.010)	(0.006)
[1,4]	0.206	0.239	0.215	0.104	0.009	[1,4]	0.219	0.226	0.221	0.127	0.011
	(0.015)	(0.018)	(0.017)	(0.014)	(0.005)		(0.011)	(0.012)	(0.011)	(0.010)	(0.003)
[4,5]	0.189	0.267	0.218	0.117	0.009	[4,5]	0.176	0.266	0.202	0.115	0.010
	(0.015)	(0.018)	(0.017)	(0.014)	(0.004)		(0.010)	(0.011)	(0.011)	(0.009)	(0.003)
[5,6]	0.158	0.215	0.141	0.102	0.006	[5,6]	0.145	0.231	0.206	0.112	0.008
	(0.015)	(0.017)	(0.016)	(0.014)	(0.004)		(0.011)	(0.011)	(0.011)	(0.010)	(0.003)
[6,7]	0.248	0.284	0.256	0.147	0.011	[6,7]	0.241	0.330	0.267	0.175	0.014
	(0.018)	(0.018)	(0.018)	(0.015)	(0.005)		(0.011)	(0.014)	(0.012)	(0.010)	(0.004)

Table 3. Downward Mobility Measures: Percentile Weight

	Q1	Q2	Q3	Q4	Q5		Q1	Q2	Q3	Q4	Q5
I. Full Sa	ample										
[0,7]	0.037	0.234	0.418	0.584	0.741						
	(0.005)	(0.009)	(0.011)	(0.011)	(0.010)						
[0,1]	0.042	0.251	0.408	0.548	0.700						
	(0.005)	(0.010)	(0.012)	(0.011)	(0.010)						
[1,7]	0.062	0.287	0.374	0.479	0.390						
	(0.005)	(0.011)	(0.011)	(0.011)	(0.010)						
[1,4]	0.022	0.169	0.250	0.280	0.108						
	(0.004)	(0.009)	(0.011)	(0.011)	(0.008)						
[4,5]	0.023	0.166	0.283	0.289	0.091						
	(0.004)	(0.009)	(0.010)	(0.011)	(0.008)						
[5,6]	0.016	0.134	0.228	0.191	0.067						
	(0.003)	(0.009)	(0.010)	(0.010)	(0.007)						
[6,7]	0.045	0.219	0.305	0.295	0.140						
	(0.005)	(0.009)	(0.011)	(0.011)	(0.008)						
II. White	.					III. No	n-White				
[0,7]	0.039	0.242	0.428	0.579	0.742	[0,7]	0.046	0.225	0.445	0.571	0.740
[0,7]	(0.006)	(0.013)	(0.015)	(0.015)	(0.013)	[0,7]	(0.008)	(0.016)	(0.018)	(0.019)	(0.017)
[0,1]	0.048	0.249	0.398	0.538	0.695	[0,1]	0.034	0.241	0.407	0.526	0.700
[0,1]	(0.007)	(0.012)	(0.015)	(0.013)	(0.012)	[0,1]	(0.007)	(0.017)	(0.019)	(0.020)	(0.019)
[1,7]	0.066	0.281	0.389	0.501	0.400	[1,7]	0.056	0.291	0.345	0.467	0.364
[1,7]	(0.006)	(0.014)	(0.014)	(0.015)	(0.014)	[1,7]	(0.008)	(0.017)	(0.018)	(0.018)	(0.018)
[1,4]	0.023	0.164	0.274	0.295	0.107	[1,4]	0.020	0.170	0.220	0.265	0.104
L - 3-1	(0.005)	(0.013)	(0.014)	(0.014)	(0.009)	[-,•]	(0.005)	(0.015)	(0.020)	(0.019)	(0.013)
[4,5]	0.022	0.163	0.295	0.304	0.116	[4,5]	0.026	0.171	0.273	0.245	0.067
[1,0]	(0.004)	(0.009)	(0.014)	(0.014)	(0.011)	[1,0]	(0.007)	(0.015)	(0.019)	(0.023)	(0.011)
[5,6]	0.014	0.138	0.239	0.203	0.078	[5,6]	0.018	0.134	0.194	0.169	0.046
[2,0]	(0.004)	(0.011)	(0.013)	(0.013)	(0.009)	[2,0]	(0.006)	(0.016)	(0.015)	(0.017)	(0.009)
[6,7]	0.045	0.229	0.307	0.310	0.148	[6,7]	0.028	0.219	0.286	0.263	0.135
[0,/]	(0.006)	(0.012)	(0.014)	(0.013)	(0.012)	[0,7]	(0.008)	(0.017)	(0.017)	(0.017)	(0.013)
	(0.000)	(0.012)	(0.014)	(0.013)	(0.012)		(0.008)	(0.017)	(0.017)	(0.017)	(0.013)

Notes: Period 0 = birth; Period 1 = fall kindergarten; Period 4 = spring 1st grade; Period 5 = spring 3rd grade; Period 6 = spring 5th grade; and, Period 7 = spring 8th grade. Q1 - Q5 refers to the first through fifth quantiles in the initial period. Bootstrap standard errors based on 250 repetitions (clustered at the child-level) in parentheses. N = 8370 (full sample), 5330 (white sample), 3040 (non-white sample), 4230 (male sample), 4140 (female sample), 2660 (low SES sample), and 5710 (high SES sample) with sample sizes rounded to the nearest 10 per NCES restricted data regulations. See text for definition of mobility measures ($\delta = 0.10$).

Table 3 (cont.). Downward Mobility Measures: Percentile Weight

	Q1	Q2	Q3	Q4	Q5		Q1	Q2	Q3	Q4	Q5
		-			-						
IV. Male	!					V. Fer	nale				
[0,7]	0.037	0.246	0.435	0.587	0.754	[0,7]	0.059	0.235	0.418	0.587	0.736
	(0.007)	(0.015)	(0.017)	(0.016)	(0.013)		(0.009)	(0.014)	(0.016)	(0.016)	(0.016)
[0,1]	0.042	0.267	0.409	0.554	0.703	[0,1]	0.061	0.256	0.408	0.541	0.702
	(0.007)	(0.016)	(0.017)	(0.017)	(0.014)		(0.010)	(0.014)	(0.017)	(0.020)	(0.015)
[1,7]	0.065	0.279	0.383	0.498	0.388	[1,7]	0.054	0.279	0.365	0.464	0.395
	(0.007)	(0.014)	(0.015)	(0.016)	(0.016)		(0.008)	(0.014)	(0.015)	(0.018)	(0.016)
[1,4]	0.020	0.165	0.260	0.304	0.125	[1,4]	0.029	0.178	0.240	0.259	0.092
	(0.005)	(0.015)	(0.013)	(0.015)	(0.012)		(0.006)	(0.013)	(0.017)	(0.016)	(0.012)
[4,5]	0.020	0.171	0.294	0.301	0.097	[4,5]	0.022	0.170	0.272	0.281	0.086
	(0.006)	(0.014)	(0.016)	(0.019)	(0.011)		(0.005)	(0.013)	(0.016)	(0.017)	(0.011)
[5,6]	0.009	0.119	0.234	0.169	0.065	[5,6]	0.023	0.161	0.219	0.194	0.074
	(0.004)	(0.012)	(0.015)	(0.013)	(0.009)		(0.006)	(0.014)	(0.015)	(0.014)	(0.009)
[6,7]	0.026	0.233	0.306	0.275	0.107	[6,7]	0.039	0.222	0.308	0.299	0.183
	(0.006)	(0.015)	(0.016)	(0.015)	(0.011)		(0.006)	(0.014)	(0.016)	(0.015)	(0.012)
VI. Low							igh SES				
[0,7]	0.049	0.244	0.447	0.541	0.748	[0,7]	0.035	0.229	0.418	0.581	0.736
	(0.009)	(0.019)	(0.020)	(0.020)	(0.017)		(0.006)	(0.012)	(0.014)	(0.014)	(0.013)
[0,1]	0.038	0.248	0.418	0.512	0.689	[0,1]	0.046	0.249	0.383	0.566	0.705
	(0.008)	(0.018)	(0.021)	(0.020)	(0.018)		(0.007)	(0.013)	(0.015)	(0.014)	(0.013)
[1,7]	0.058	0.279	0.374	0.475	0.367	[1,7]	0.063	0.284	0.375	0.466	0.396
	(0.009)	(0.018)	(0.020)	(0.021)	(0.019)		(0.006)	(0.012)	(0.013)	(0.014)	(0.012)
[1,4]	0.019	0.173	0.245	0.271	0.106	[1,4]	0.023	0.167	0.253	0.288	0.109
	(0.006)	(0.017)	(0.021)	(0.024)	(0.014)		(0.005)	(0.011)	(0.013)	(0.013)	(0.010)
[4,5]	0.023	0.183	0.276	0.277	0.066	[4,5]	0.024	0.167	0.279	0.289	0.103
	(0.007)	(0.020)	(0.020)	(0.019)	(0.016)		(0.004)	(0.010)	(0.013)	(0.014)	(0.010)
[5,6]	0.017	0.124	0.215	0.166	0.062	[5,6]	0.015	0.140	0.225	0.200	0.073
	(0.006)	(0.016)	(0.018)	(0.016)	(0.012)		(0.004)	(0.010)	(0.012)	(0.011)	(0.008)
[6,7]	0.028	0.235	0.299	0.286	0.128	[6,7]	0.046	0.221	0.314	0.300	0.140
	(0.008)	(0.017)	(0.017)	(0.018)	(0.015)		(0.006)	(0.011)	(0.012)	(0.013)	(0.010)

Table 4. Mobility Measures: Percentile Height

1 able 4.	M(P)	M(ρ)	M _S	CF(0)	CF _R (0)		M(P)	Μ(ρ)	$M_{\rm S}$	CF(0)	CF _R (0)
		141(b)	IVIS	CF(U)	CFR(U)		MI(F)	141(h)	IVIS	CF(U)	Cr _R (U)
I. Full S	-										
[1,7]	0.180	0.287	0.207	0.111	0.130						
	(0.002)	(0.006)	(0.006)	(0.002)	(0.003)						
[1,4]	0.087	0.091	0.062	0.035	0.041						
	(0.001)	(0.003)	(0.002)	(0.002)	(0.002)						
[4,5]	0.082	0.080	0.072	0.031	0.034						
	(0.001)	(0.003)	(0.004)	(0.001)	(0.001)						
[5,6]	0.090	0.088	0.088	0.040	0.045						
	(0.001)	(0.003)	(0.005)	(0.002)	(0.002)						
[6,7]	0.153	0.211	0.172	0.074	0.092						
	(0.002)	(0.005)	(0.005)	(0.002)	(0.003)						
II. Whit	te					III. Noi	n-White				
[1,7]	0.180	0.285	0.203	0.104	0.130	[1,7]	0.180	0.285	0.211	0.120	0.128
· -	(0.003)	(0.008)	(0.008)	(0.003)	(0.004)	, -	(0.003)	(0.010)	(0.009)	(0.004)	(0.005)
[1,4]	0.085	0.089	0.058	0.033	0.039	[1,4]	0.089	0.094	0.068	0.040	0.044
	(0.002)	(0.004)	(0.002)	(0.002)	(0.002)		(0.002)	(0.005)	(0.004)	(0.003)	(0.003)
[4,5]	0.078	0.072	0.066	0.026	0.030	[4,5]	0.089	0.093	0.082	0.039	0.040
- / -	(0.002)	(0.003)	(0.005)	(0.001)	(0.001)	2 / 2	(0.002)	(0.005)	(0.007)	(0.002)	(0.002)
[5,6]	0.087	0.083	0.081	0.033	0.040	[5,6]	0.094	0.096	0.099	0.053	0.052
, -	(0.002)	(0.004)	(0.006)	(0.003)	(0.003)	- , -	(0.002)	(0.005)	(0.008)	(0.005)	(0.004)
[6,7]	0.147	0.192	0.156	0.064	0.085	[6,7]	0.162	0.235	0.190	0.091	0.100
· / -	(0.002)	(0.006)	(0.006)	(0.002)	(0.003)	- / -	(0.003)	(0.009)	(0.008)	(0.003)	(0.004)
IV. Mal	le					V. Fem	ale				
[1,7]	0.180	0.282	0.190	0.109	0.129	[1,7]	0.177	0.283	0.221	0.110	0.127
L - 9/J	(0.003)	(0.009)	(0.008)	(0.004)	(0.004)	L * 9/J	(0.003)	(0.008)	(0.007)	(0.004)	(0.004)
[1,4]	0.082	0.088	0.057	0.034	0.041	[1,4]	0.086	0.089	0.065	0.036	0.039
L-7"J	(0.002)	(0.005)	(0.004)	(0.003)	(0.003)	L-9-1	(0.002)	(0.005)	(0.003)	(0.002)	(0.002)
[4,5]	0.078	0.003)	0.067	0.027	0.032	[4,5]	0.085	0.084	0.076	0.034	0.035
[-,~]	(0.002)	(0.004)	(0.005)	(0.027)	(0.002)	[- 1 ,~]	(0.002)	(0.004)	(0.005)	(0.002)	(0.002)
[5,6]	0.076	0.070	0.079	0.032	0.037	[5,6]	0.102	0.104	0.003)	0.048	0.051
[~,0]	(0.002)	(0.004)	(0.006)	(0.003)	(0.003)	[~,0]	(0.002)	(0.004)	(0.006)	(0.003)	(0.003)
[6,7]	0.131	0.150	0.138	0.055	0.069	[6,7]	0.175	0.259	0.199	0.092	0.111
r^,,1	(0.002)	(0.005)	(0.006)	(0.002)	(0.003)	[~,·]	(0.003)	(0.008)	(0.008)	(0.003)	(0.004)
T/T -	, ,	(3.302)	(2.300)	(3.30-)	(3.000)	****	, ,	(3.300)	(3.300)	(3.300)	(5.00.)
VI. Low		0.205	0.210	0.120	0.121	VII. Hi	_	0.202	0.201	0.102	0.121
[1,7]	0.182	0.295	0.218	0.128	0.131	[1,7]	0.180	0.283	0.201	0.103	0.131
F4 43	(0.004)	(0.011)	(0.011)	(0.005)	(0.005)	F4 43	(0.002)	(0.007)	(0.007)	(0.003)	(0.004)
[1,4]	0.088	0.095	0.065	0.041	0.042	[1,4]	0.086	0.089	0.061	0.033	0.041
F.4. = 3	(0.002)	(0.006)	(0.004)	(0.003)	(0.003)	F4 =3	(0.002)	(0.004)	(0.003)	(0.002)	(0.003)
[4,5]	0.082	0.078	0.073	0.035	0.033	[4,5]	0.082	0.081	0.072	0.029	0.034
	(0.002)	(0.005)	(0.007)	(0.002)	(0.002)		(0.002)	(0.004)	(0.005)	(0.001)	(0.002)
[5,6]	0.087	0.080	0.090	0.043	0.040	[5,6]	0.091	0.092	0.087	0.039	0.048
	(0.002)	(0.005)	(0.007)	(0.003)	(0.003)		(0.002)	(0.004)	(0.006)	(0.003)	(0.003)
[6,7]	0.155	0.219	0.180	0.088	0.094	[6,7]	0.152	0.206	0.167	0.068	0.091
	(0.003)	(0.009)	(0.008)	(0.004)	(0.004)		(0.002)	(0.006)	(0.006)	(0.002)	(0.003)

Notes: See Table 1 for details.

Table 5. Upward Mobility Measures: Percentile Height

	Q1	Q2	Q3	Q4	Q5		Q1	Q2	Q3	Q4	Q5
I. Full Sa											
[1,7]	0.379	0.426	0.342	0.255	0.044						
- / -	(0.011)	(0.011)	(0.011)	(0.010)	(0.005)						
[1,4]	0.107	0.187	0.176	0.122	0.013						
L-, -J	(0.008)	(0.009)	(0.009)	(0.009)	(0.003)						
[4,5]	0.091	0.198	0.190	0.104	0.011						
[-90]	(0.007)	(0.009)	(0.011)	(0.008)	(0.003)						
[5,6]	0.141	0.202	0.198	0.136	0.015						
[5,0]	(0.008)	(0.009)	(0.009)	(0.008)	(0.003)						
[6,7]	0.304	0.404	0.355	0.260	0.044						
[0,7]	(0.011)	(0.010)	(0.010)	(0.010)	(0.005)						
	(0.011)	(0.010)	(0.010)	(0.010)	(0.003)						
II. White						III. Noi					
[1,7]	0.388	0.417	0.343	0.260	0.044	[1,7]	0.371	0.413	0.369	0.227	0.038
	(0.014)	(0.014)	(0.014)	(0.012)	(0.006)		(0.018)	(0.020)	(0.018)	(0.017)	(0.008)
[1,4]	0.102	0.182	0.178	0.130	0.017	[1,4]	0.125	0.189	0.175	0.112	0.010
	(0.009)	(0.011)	(0.012)	(0.010)	(0.004)		(0.013)	(0.015)	(0.016)	(0.013)	(0.004)
[4,5]	0.080	0.189	0.176	0.096	0.009	[4,5]	0.120	0.199	0.216	0.129	0.013
	(0.008)	(0.010)	(0.012)	(0.009)	(0.003)		(0.013)	(0.015)	(0.015)	(0.014)	(0.005)
[5,6]	0.132	0.179	0.225	0.132	0.012	[5,6]	0.156	0.236	0.189	0.116	0.021
	(0.010)	(0.012)	(0.012)	(0.011)	(0.004)		(0.015)	(0.017)	(0.015)	(0.012)	(0.006)
[6,7]	0.287	0.379	0.344	0.251	0.039	[6,7]	0.312	0.447	0.347	0.253	0.045
	(0.013)	(0.014)	(0.014)	(0.012)	(0.006)		(0.018)	(0.020)	(0.018)	(0.018)	(0.008)
IV. Male	;					V. Fem	ale				
[1,7]	0.419	0.418	0.353	0.261	0.039	[1,7]	0.356	0.419	0.335	0.237	0.045
.,,	(0.015)	(0.015)	(0.016)	(0.014)	(0.007)	.,1	(0.015)	(0.016)	(0.015)	(0.014)	(0.007)
[1,4]	0.106	0.176	0.163	0.105	0.014	[1,4]	0.104	0.215	0.186	0.117	0.015
L-, -J	(0.011)	(0.013)	(0.013)	(0.011)	(0.004)	L=, -1	(0.011)	(0.013)	(0.014)	(0.011)	(0.004)
[4,5]	0.090	0.184	0.160	0.091	0.006	[4,5]	0.092	0.212	0.191	0.125	0.019
[.,0]	(0.010)	(0.014)	(0.013)	(0.012)	(0.003)	[.,]	(0.010)	(0.014)	(0.013)	(0.012)	(0.005)
[5,6]	0.096	0.160	0.165	0.114	0.012	[5,6]	0.174	0.249	0.247	0.157	0.017
[5,0]	(0.009)	(0.012)	(0.015)	(0.011)	(0.004)	[5,0]	(0.012)	(0.013)	(0.014)	(0.012)	(0.005)
[6,7]	0.269	0.352	0.308	0.198	0.033	[6,7]	0.348	0.455	0.404	0.294	0.054
[0,7]	(0.014)	(0.015)	(0.015)	(0.013)	(0.006)	[0,7]	(0.017)	(0.017)	(0.017)	(0.015)	(0.010)
		(010-2)	(010-0)	(01010)	(31333)			(01011)	(0.01.)	(010-0)	(010-0)
VI. Low		0.407	0.245	0.265	0.051	VII. Hi	_	0.425	0.247	0.242	0.027
[1,7]	0.402	0.407	0.345	0.265	0.051	[1,7]	0.396	0.425	0.347	0.242	0.037
	(0.018)	(0.019)	(0.019)	(0.018)	(0.009)		(0.013)	(0.013)	(0.011)	(0.012)	(0.005)
[1,4]	0.114	0.185	0.170	0.117	0.009	[1,4]	0.113	0.188	0.181	0.126	0.014
F A #5	(0.014)	(0.017)	(0.016)	(0.016)	(0.004)	F.4.==	(0.010)	(0.011)	(0.012)	(0.010)	(0.004)
[4,5]	0.101	0.210	0.199	0.130	0.023	[4,5]	0.089	0.196	0.189	0.088	0.011
	(0.014)	(0.018)	(0.017)	(0.016)	(0.007)		(0.008)	(0.011)	(0.011)	(0.009)	(0.003)
[5,6]	0.160	0.204	0.173	0.134	0.011	[5,6]	0.131	0.208	0.221	0.124	0.018
	(0.015)	(0.017)	(0.017)	(0.014)	(0.005)		(0.009)	(0.011)	(0.011)	(0.010)	(0.004)
		0.400		0.070							
[6,7]	0.305	0.422	0.339	0.252	0.055	[6,7]	0.302	0.402	0.354	0.262	0.038

Notes: See Table 2.

Table 6. Downward Mobility Measures: Percentile Height

	Q1	Q2	Q3	Q4	Q5		Q1	Q2	Q3	Q4	Q5
I. Full S	Sample										
[1,7]	0.042	0.222	0.354	0.385	0.398						
	(0.005)	(0.010)	(0.012)	(0.011)	(0.011)						
[1,4]	0.011	0.114	0.180	0.185	0.134						
	(0.003)	(0.008)	(0.010)	(0.010)	(0.007)						
[4,5]	0.011	0.086	0.166	0.178	0.135						
	(0.003)	(0.007)	(0.010)	(0.009)	(0.008)						
[5,6]	0.011	0.131	0.212	0.215	0.118						
	(0.003)	(0.009)	(0.009)	(0.010)	(0.008)						
[6,7]	0.024	0.195	0.303	0.333	0.333						
	(0.004)	(0.008)	(0.012)	(0.011)	(0.010)						
II. Whit	te					III. No	n-White				
[1,7]	0.046	0.236	0.352	0.376	0.405	[1,7]	0.041	0.220	0.356	0.402	0.388
L - 9, J	(0.006)	(0.013)	(0.015)	(0.013)	(0.013)	[-,,]	(0.007)	(0.016)	(0.019)	(0.022)	(0.018)
[1,4]	0.010	0.104	0.169	0.171	0.141	[1,4]	0.012	0.126	0.211	0.205	0.130
L - 9-J	(0.003)	(0.010)	(0.013)	(0.011)	(0.010)	L - 7-J	(0.006)	(0.014)	(0.015)	(0.020)	(0.013)
[4,5]	0.010	0.096	0.168	0.170	0.117	[4,5]	0.015	0.084	0.168	0.194	0.149
[1,0]	(0.003)	(0.009)	(0.012)	(0.012)	(0.009)	[.,e]	(0.005)	(0.012)	(0.015)	(0.016)	(0.014)
[5,6]	0.009	0.131	0.164	0.218	0.117	[5,6]	0.018	0.128	0.257	0.226	0.125
[0,0]	(0.004)	(0.011)	(0.014)	(0.014)	(0.010)	[0,0]	(0.006)	(0.015)	(0.017)	(0.018)	(0.015)
[6,7]	0.024	0.208	0.312	0.333	0.317	[6,7]	0.028	0.177	0.314	0.348	0.359
[0,1]	(0.005)	(0.012)	(0.015)	(0.014)	(0.012)	[0,,,]	(0.007)	(0.014)	(0.017)	(0.017)	(0.017)
IV. Mal						V. Fem	nala				
[1,7]	0.048	0.253	0.341	0.378	0.388	[1,7]	0.040	0.207	0.350	0.370	0.395
[1,/]	(0.008)	(0.013)	(0.015)	(0.017)	(0.014)	[1,/]	(0.006)	(0.013)	(0.015)	(0.017)	(0.014)
[1,4]	0.008)	0.013)	0.152	0.163	0.139	[1,4]	0.011	0.106	0.182	0.185	0.132
[1,7]	(0.003)	(0.010)	(0.015)	(0.015)	(0.011)	[1,7]	(0.004)	(0.010)	(0.013)	(0.015)	(0.011)
[4,5]	0.012	0.010)	0.155	0.154	0.117	[4,5]	0.004)	0.010)	0.170	0.188	0.140
[4,3]	(0.003)	(0.010)	(0.013)	(0.012)	(0.011)	[4,3]	(0.004)	(0.010)	(0.014)	(0.013)	(0.011)
[5 6]	0.003)	0.110	0.159	0.168	0.011)	[5,6]	0.013	0.166	0.242	0.265	0.164
[5,6]	(0.003)	(0.012)	(0.014)	(0.013)	(0.009)	[3,0]	(0.004)	(0.013)	(0.014)	(0.016)	(0.013)
[6,7]	0.025	0.211	0.342	0.342	0.243	[6,7]	0.022	0.207	0.308	0.379	0.432
[0,7]	(0.006)	(0.014)	(0.015)	(0.017)	(0.014)	[0,7]	(0.005)	(0.014)	(0.015)	(0.014)	(0.016)
	(0.000)	(0.014)	(0.013)	(0.017)	(0.014)		(0.003)	(0.014)	(0.013)	(0.014)	(0.010)
VI. Low	v SES					VII. H	igh SES				
[1,7]	0.036	0.209	0.335	0.359	0.421	[1,7]	0.045	0.233	0.350	0.399	0.385
	(0.008)	(0.017)	(0.021)	(0.022)	(0.020)		(0.007)	(0.012)	(0.013)	(0.013)	(0.013)
[1,4]	0.013	0.124	0.173	0.184	0.153	[1,4]	0.013	0.103	0.186	0.191	0.124
	(0.007)	(0.015)	(0.016)	(0.016)	(0.015)		(0.004)	(0.010)	(0.012)	(0.012)	(0.009)
[4,5]	0.008	0.100	0.152	0.168	0.134	[4,5]	0.008	0.089	0.176	0.176	0.133
	(0.004)	(0.014)	(0.016)	(0.017)	(0.014)		(0.003)	(0.008)	(0.011)	(0.011)	(0.009)
[5,6]	0.011	0.151	0.205	0.199	0.126	[5,6]	0.010	0.137	0.198	0.233	0.117
	(0.005)	(0.017)	(0.018)	(0.018)	(0.014)		(0.003)	(0.010)	(0.013)	(0.013)	(0.009)
[6,7]	0.021	0.193	0.297	0.335	0.338	[6,7]	0.026	0.201	0.303	0.335	0.330
	(0.007)	(0.016)	(0.016)	(0.019)	(0.016)		(0.005)	(0.012)	(0.013)	(0.013)	(0.012)

Notes: See Table 3.

Table 7. Mobility Measures: Percentile BMI

	M(P)	Μ(ρ)	M_S	CF(0)	CF _R (0)		M(P)	Μ(ρ)	M_{S}	CF(0)	CF _R (0)
I. Full S	Sample										
[1,7]	0.217	0.405	0.498	0.358	0.243						
L-9- J	(0.003)	(0.008)	(0.009)	(0.021)	(0.005)						
[1,4]	0.147	0.218	0.437	0.221	0.138						
L - 9•J	(0.002)	(0.006)	(0.011)	(0.013)	(0.004)						
[4,5]	0.125	0.155	0.313	0.113	0.101						
L - , - J	(0.002)	(0.004)	(0.017)	(0.010)	(0.004)						
[5,6]	0.094	0.086	0.170	0.042	0.051						
[-,0]	(0.001)	(0.003)	(0.011)	(0.003)	(0.002)						
[6,7]	0.118	0.126	0.230	0.050	0.067						
[0,7]	(0.002)	(0.003)	(0.011)	(0.002)	(0.002)						
TT 3371 **	, ,	(,	(,	(,	(/	TIT N	11 71 • 4				
II. Whit		0.407	0.402	0.200	0.255	III. Noi		0.276	0.510	0.206	0.224
[1,7]	0.224	0.427	0.493	0.390	0.255	[1,7]	0.207	0.376	0.510	0.306	0.224
F4 43	(0.003)	(0.011)	(0.011)	(0.028)	(0.008)	F4 43	(0.004)	(0.012)	(0.018)	(0.027)	(0.008)
[1,4]	0.151	0.231	0.435	0.234	0.144	[1,4]	0.140	0.199	0.443	0.199	0.128
F.4. #3	(0.003)	(0.007)	(0.014)	(0.017)	(0.005)	5.4 # 3	(0.003)	(0.009)	(0.021)	(0.018)	(0.006)
[4,5]	0.129	0.167	0.307	0.113	0.105	[4,5]	0.119	0.136	0.326	0.113	0.092
	(0.002)	(0.006)	(0.024)	(0.013)	(0.005)		(0.003)	(0.006)	(0.026)	(0.017)	(0.005)
[5,6]	0.097	0.091	0.160	0.042	0.053	[5,6]	0.091	0.080	0.193	0.042	0.047
	(0.002)	(0.003)	(0.012)	(0.004)	(0.002)		(0.002)	(0.004)	(0.023)	(0.006)	(0.003)
[6,7]	0.122	0.137	0.227	0.053	0.072	[6,7]	0.111	0.112	0.237	0.045	0.060
	(0.002)	(0.004)	(0.013)	(0.002)	(0.003)		(0.002)	(0.005)	(0.021)	(0.003)	(0.003)
IV. Mal	le					V. Fem	ale				
[1,7]	0.221	0.416	0.532	0.408	0.250	[1,7]	0.212	0.391	0.458	0.307	0.233
	(0.003)	(0.012)	(0.014)	(0.035)	(0.008)		(0.003)	(0.011)	(0.010)	(0.023)	(0.007)
[1,4]	0.152	0.233	0.467	0.245	0.145	[1,4]	0.142	0.203	0.403	0.195	0.130
	(0.003)	(0.009)	(0.018)	(0.022)	(0.006)		(0.003)	(0.007)	(0.013)	(0.013)	(0.005)
[4,5]	0.129	0.167	0.340	0.129	0.108	[4,5]	0.121	0.143	0.287	0.096	0.093
	(0.002)	(0.006)	(0.029)	(0.018)	(0.005)		(0.002)	(0.006)	(0.020)	(0.009)	(0.004)
[5,6]	0.094	0.087	0.180	0.039	0.052	[5,6]	0.095	0.085	0.162	0.044	0.050
	(0.002)	(0.004)	(0.015)	(0.004)	(0.003)		(0.002)	(0.003)	(0.017)	(0.006)	(0.003)
[6,7]	0.113	0.120	0.243	0.048	0.064	[6,7]	0.120	0.128	0.210	0.052	0.068
	(0.002)	(0.004)	(0.019)	(0.002)	(0.003)		(0.002)	(0.005)	(0.011)	(0.002)	(0.003)
VI. Low	v SES					VII. Hi	gh SES				
[1,7]	0.214	0.389	0.501	0.334	0.234	[1,7]	0.218	0.414	0.499	0.370	0.246
.,,	(0.004)	(0.014)	(0.017)	(0.032)	(0.010)	.,.	(0.003)	(0.010)	(0.011)	(0.027)	(0.007)
[1,4]	0.144	0.211	0.428	0.201	0.132	[1,4]	0.149	0.223	0.442	0.230	0.140
L - 7 •J	(0.003)	(0.010)	(0.018)	(0.017)	(0.007)	L-9 ·J	(0.003)	(0.008)	(0.014)	(0.016)	(0.005)
[4,5]	0.123	0.145	0.318	0.124	0.100	[4,5]	0.127	0.160	0.313	0.107	0.100
L -7~ J	(0.003)	(0.007)	(0.031)	(0.021)	(0.006)	L -9~ J	(0.002)	(0.006)	(0.021)	(0.012)	(0.004)
[5,6]	0.090	0.007)	0.144	0.021)	0.044	[5,6]	0.002)	0.000)	0.182	0.012)	0.054
[2,0]	(0.002)	(0.004)	(0.012)	(0.003)	(0.003)	[5,0]	(0.002)	(0.093)	(0.015)	(0.005)	(0.002)
[6,7]	0.002)	0.004)	0.248	0.043	0.062	[6,7]	0.121	0.135	0.224	0.054	0.002)
[Մ,/]						[0,/]					
	(0.003)	(0.005)	(0.025)	(0.003)	(0.003)		(0.002)	(0.004)	(0.012)	(0.002)	(0.002)

Notes: See Table 1 for details.

Table 8. Upward Mobility Measures: Percentile BMI

	Q1	Q2	Q3	Q4	Q5		Q1	Q2	Q3	Q4	Q5
I. Full Sa	ample										
[1,7]	0.543	0.409	0.311	0.189	0.032						
	(0.011)	(0.012)	(0.011)	(0.009)	(0.004)						
[1,4]	0.380	0.315	0.244	0.146	0.014						
	(0.010)	(0.010)	(0.010)	(0.008)	(0.003)						
[4,5]	0.329	0.274	0.225	0.118	0.013						
	(0.010)	(0.010)	(0.009)	(0.007)	(0.003)						
[5,6]	0.224	0.235	0.196	0.115	0.010						
- , -	(0.009)	(0.009)	(0.010)	(0.008)	(0.003)						
[6,7]	0.292	0.319	0.256	0.160	0.016						
- , -	(0.011)	(0.011)	(0.010)	(0.009)	(0.003)						
II \\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\	, ,	, ,	, ,	, ,	, ,	III Nov	. White				
II. White		0.402	0.202	0.106	0.024	III. Noi		0.200	0.212	0.172	0.022
[1,7]	0.560	0.402	0.302	0.196	0.034	[1,7]	0.521	0.399	0.312	0.173	0.033
F1 41	(0.014)	(0.013)	(0.013)	(0.011)	(0.005)	F1 41	(0.017)	(0.019)	(0.017)	(0.016)	(0.007)
[1,4]	0.386	0.320	0.230	0.160	0.017	[1,4]	0.363	0.312	0.259	0.130	0.008
	(0.012)	(0.014)	(0.012)	(0.010)	(0.004)		(0.016)	(0.017)	(0.016)	(0.014)	(0.005)
[4,5]	0.324	0.284	0.237	0.134	0.009	[4,5]	0.333	0.273	0.209	0.107	0.010
	(0.012)	(0.014)	(0.013)	(0.010)	(0.003)		(0.017)	(0.017)	(0.014)	(0.011)	(0.004)
[5,6]	0.229	0.240	0.206	0.115	0.011	[5,6]	0.225	0.222	0.163	0.117	0.012
	(0.010)	(0.013)	(0.013)	(0.010)	(0.004)		(0.016)	(0.016)	(0.014)	(0.013)	(0.005)
[6,7]	0.303	0.333	0.271	0.156	0.022	[6,7]	0.291	0.292	0.245	0.143	0.015
	(0.013)	(0.013)	(0.013)	(0.011)	(0.005)		(0.017)	(0.017)	(0.016)	(0.014)	(0.005)
IV. Male	e					V. Fem	ale				
[1,7]	0.557	0.389	0.319	0.183	0.024	[1,7]	0.517	0.423	0.307	0.192	0.034
	(0.014)	(0.014)	(0.015)	(0.014)	(0.005)		(0.016)	(0.016)	(0.015)	(0.013)	(0.006)
[1,4]	0.405	0.323	0.251	0.148	0.011	[1,4]	0.356	0.313	0.232	0.156	0.016
	(0.014)	(0.015)	(0.014)	(0.012)	(0.004)		(0.014)	(0.015)	(0.014)	(0.014)	(0.005)
[4,5]	0.321	0.288	0.213	0.117	0.015	[4,5]	0.324	0.259	0.233	0.120	0.011
	(0.013)	(0.014)	(0.013)	(0.011)	(0.004)		(0.014)	(0.014)	(0.013)	(0.011)	(0.004)
[5,6]	0.229	0.219	0.183	0.114	0.013	[5,6]	0.221	0.250	0.205	0.111	0.004
	(0.013)	(0.014)	(0.014)	(0.012)	(0.004)		(0.014)	(0.014)	(0.015)	(0.012)	(0.003)
[6,7]	0.275	0.322	0.251	0.137	0.014	[6,7]	0.291	0.324	0.266	0.164	0.016
	(0.014)	(0.014)	(0.014)	(0.012)	(0.004)		(0.015)	(0.015)	(0.015)	(0.012)	(0.005)
VI. Low	SES					VII. Hi	gh SES				
[1,7]	0.540	0.392	0.329	0.177	0.034	[1,7]	0.540	0.414	0.291	0.191	0.032
	(0.019)	(0.020)	(0.020)	(0.016)	(0.007)		(0.013)	(0.013)	(0.013)	(0.011)	(0.005)
[1,4]	0.378	0.307	0.248	0.138	0.008	[1,4]	0.379	0.320	0.232	0.156	0.018
- / -	(0.015)	(0.019)	(0.018)	(0.014)	(0.004)	- / -	(0.012)	(0.013)	(0.012)	(0.011)	(0.004)
[4,5]	0.340	0.281	0.209	0.111	0.011	[4,5]	0.326	0.290	0.232	0.126	0.011
L /- J	(0.016)	(0.017)	(0.016)	(0.012)	(0.005)	L)- J	(0.012)	(0.014)	(0.011)	(0.010)	(0.003)
[5,6]	0.205	0.222	0.196	0.119	0.011	[5,6]	0.236	0.256	0.204	0.109	0.013
[~,~]	(0.018)	(0.019)	(0.016)	(0.014)	(0.005)	[~] ~]	(0.011)	(0.012)	(0.012)	(0.010)	(0.003)
[6,7]	0.288	0.294	0.243	0.179	0.013	[6,7]	0.300	0.326	0.277	0.155	0.012
[0,1]	(0.020)	(0.018)	(0.018)	(0.016)	(0.005)	[0,1]	(0.012)	(0.013)	(0.013)	(0.010)	(0.004)
	(0.020)	(0.010)	(0.010)	(0.010)	(0.003)		(0.012)	(0.013)	(0.013)	(0.010)	(0.007)

Notes: See Table 2.

Table 9. Downward Mobility Measures: Percentile BMI

	Q1	Q2	Q3	Q4	Q5		Q1	Q2	Q3	Q4	Q5
I. Full S	Sample										
[1,7]	0.052	0.311	0.417	0.513	0.410						
[-9.]	(0.005)	(0.011)	(0.011)	(0.011)	(0.011)						
[1,4]	0.046	0.245	0.361	0.361	0.186						
[-,-]	(0.005)	(0.010)	(0.010)	(0.011)	(0.011)						
[4,5]	0.047	0.250	0.354	0.348	0.117						
L - / - J	(0.005)	(0.010)	(0.012)	(0.011)	(0.008)						
[5,6]	0.035	0.206	0.253	0.222	0.090						
[-,-]	(0.004)	(0.010)	(0.011)	(0.010)	(0.007)						
[6,7]	0.038	0.228	0.313	0.306	0.142						
[-,-]	(0.005)	(0.010)	(0.010)	(0.010)	(0.009)						
** ***		(333-3)	(0.0-0)	(000-0)	(0100)	*** **	****				
II. Whit							n-White				0.4.0
[1,7]	0.046	0.325	0.431	0.514	0.445	[1,7]	0.046	0.294	0.406	0.525	0.368
	(0.007)	(0.014)	(0.014)	(0.013)	(0.013)		(0.009)	(0.018)	(0.020)	(0.020)	(0.018)
[1,4]	0.048	0.251	0.365	0.368	0.201	[1,4]	0.041	0.235	0.348	0.365	0.165
	(0.006)	(0.013)	(0.015)	(0.015)	(0.012)		(0.009)	(0.018)	(0.016)	(0.017)	(0.016)
[4,5]	0.038	0.250	0.337	0.354	0.162	[4,5]	0.061	0.268	0.339	0.328	0.089
	(0.006)	(0.014)	(0.014)	(0.014)	(0.013)		(0.009)	(0.018)	(0.018)	(0.019)	(0.013)
[5,6]	0.034	0.207	0.266	0.227	0.101	[5,6]	0.035	0.199	0.253	0.189	0.087
	(0.005)	(0.012)	(0.012)	(0.013)	(0.010)		(0.007)	(0.016)	(0.018)	(0.016)	(0.012)
[6,7]	0.038	0.234	0.323	0.310	0.161	[6,7]	0.048	0.253	0.294	0.259	0.127
	(0.006)	(0.012)	(0.013)	(0.013)	(0.010)		(0.010)	(0.016)	(0.019)	(0.017)	(0.015)
IV. Mal	le					V. Fem	nale				
[1,7]	0.053	0.322	0.424	0.520	0.411	[1,7]	0.054	0.295	0.407	0.506	0.393
	(0.007)	(0.014)	(0.016)	(0.015)	(0.016)		(0.008)	(0.014)	(0.017)	(0.017)	(0.016)
[1,4]	0.047	0.259	0.359	0.374	0.193	[1,4]	0.051	0.238	0.354	0.350	0.184
	(0.008)	(0.014)	(0.015)	(0.016)	(0.014)		(0.008)	(0.013)	(0.015)	(0.017)	(0.015)
[4,5]	0.052	0.249	0.383	0.359	0.116	[4,5]	0.045	0.237	0.339	0.326	0.115
	(0.007)	(0.015)	(0.016)	(0.017)	(0.011)		(0.007)	(0.015)	(0.015)	(0.015)	(0.013)
[5,6]	0.037	0.199	0.245	0.228	0.088	[5,6]	0.031	0.211	0.252	0.207	0.103
	(0.007)	(0.015)	(0.015)	(0.016)	(0.010)		(0.006)	(0.016)	(0.015)	(0.014)	(0.012)
[6,7]	0.034	0.221	0.297	0.286	0.131	[6,7]	0.044	0.230	0.318	0.331	0.147
	(0.006)	(0.014)	(0.015)	(0.017)	(0.012)		(0.008)	(0.014)	(0.015)	(0.015)	(0.013)
VI. Low	v SES					VII. Hi	ioh SES				
[1,7]	0.056	0.307	0.414	0.535	0.379	[1,7]	0.053	0.307	0.425	0.503	0.426
[=,,]	(0.011)	(0.018)	(0.020)	(0.019)	(0.020)	[2,,]	(0.007)	(0.014)	(0.015)	(0.014)	(0.013)
[1,4]	0.043	0.239	0.374	0.392	0.160	[1,4]	0.049	0.252	0.356	0.347	0.198
L±9 [∓] J	(0.011)	(0.019)	(0.020)	(0.019)	(0.015)	[* ; "]	(0.006)	(0.011)	(0.013)	(0.014)	(0.012)
[4,5]	0.041	0.283	0.355	0.343	0.013)	[4,5]	0.049	0.243	0.336	0.359	0.131
[-,-]	(0.008)	(0.022)	(0.019)	(0.019)	(0.016)	[-,-]	(0.006)	(0.014)	(0.013)	(0.013)	(0.011)
[5,6]	0.030	0.202	0.226	0.215	0.102	[5,6]	0.034	0.214	0.266	0.229	0.100
[2,0]	(0.008)	(0.018)	(0.018)	(0.017)	(0.015)	[2,0]	(0.005)	(0.012)	(0.013)	(0.014)	(0.009)
[6,7]	0.049	0.256	0.312	0.277	0.147	[6,7]	0.045	0.219	0.319	0.307	0.009)
[0,/]	(0.009)	(0.019)	(0.017)	(0.019)	(0.015)	[0,/]	(0.006)	(0.011)	(0.012)	(0.014)	(0.010)
	(0.009)	(0.019)	(0.017)	(0.019)	(0.013)		(0.000)	(0.011)	(0.012)	(0.014)	(0.010)

Notes: See Table 3.